The World Court of Justice

The Alternative to Wars, Terrorism and Politics

© 2004 by Truth and Justice Publications Ltd. *All rights reserved*.

No part of this publication may be reproduced, stored in a retrieval system, or transmitted in any form or by any means, electronic, mechanical, photocopying, recording or otherwise without prior written permission of the publisher.

The publisher makes no warranty either expressed or implied of fitness of this publication for any purpose.

Published by:

Truth and Justice Publications Ltd PO Box 10121 Birmingham B27 7YS United Kingdom

Tel: +44 (0)121 706 9614 Fax: +44 (0)121 707 9832

Web: www.truth-and-justice.co.uk

ISBN 1-904941-00-1

Date of First Publication: 4th March, 2004

Typeset, printed and bound in Great Britain by Truth and Justice Publications Ltd.

Preface

The global violence which ushered in the Third Millennium poses the questions:

"Can Mankind ever live in peace?"

"Was not all Human History just a never-ending war between tribes and nations seeking to conquer and dominate each other?"

This book identifies the root of the problem and provides the blueprint for a peaceful world.

Global problems need global solutions. The transition from the tribal partisan mindset which dominates all the present day governments will be neither instant nor painless. Nor will this tribal partisan mindset be relinquished willingly without struggle. But the relentless Force of Reality will make those in power abandon their dreams and ambitions and will impose on them its own Irresistible Will.

This is our first publication on the Principles of Global Government. It contains documents and articles relating to the World Court of Justice — an impersonal supranational institution set up on the Internet in 2001.

This book is divided into six parts.

Part I explains the reasons why the World Court of Justice is necessary.

Part II contains the Principles, Rules and Procedures of the World Court of Justice.

Part III explains the differences between the World Court of Justice and the other international courts.

Part IV illustrates Global Thinking, without which there is no hope for world peace and security.

Part V gives examples of application of Global Thinking to situations arising out of the present day lawlessness.

Part VI shows examples of interaction with the people who are at the heart of the present violence: the US Administration and Osama bin Laden.

Contents

I	Wh	y WCJ?	1
1	Why	the World Court of Justice?	3
	1.1	No Law between Nations	3
	1.2	Search for Supranational Justice	4
	1.3	Politics and Justice	4
	1.4	Composition and Procedure	5
	1.5	Validation and Execution of Judgements	5
2	Just	ice through the Ages	7
	2.1	Early Beginnings	7
	2.2	Formalization of Law	8
	2.3	Independent Courts and Supremacy of Law	8
	2.4	The Socialist Experiment	9
	2.5	The Age of Political Convenience	9
	2.6	Government without Politics	11
	2.7	The International Scene	11
II	Pr	inciples, Rules, Procedure	13
3	Fun	damental Principles of Government	15
4	Rule	es of Procedure of the World Court of Justice	19
		Purpose of the World Court of Justice	19
		2 Location	19
		3 Composition	19
		4 Jurisdiction	20
		5 Principles of Operation	20
		6 Causes of Action	21
		7 Parties	22
		8 Commencement of Actions	22
		9 Who can commence an Action?	22

iv CONTENTS

	10	Stages of Action in the World Court of Justice	22
	11	Pleadings	22
	12	Statements of Fact	23
	13	Statements of Relevance in Support of Facts	23
	14	Descriptions of Items of Evidence	23
	15	Statements of Relevance in Support of Descriptions of Items	
		of Evidence	24
	16	Burden of Proof	24
	17	Method of Proof	24
	18	Standard of Proof	25
	19	Claim	25
	20	Defence	25
	21	Reply	26
	22	Summary Judgement	26
	23	Default Judgement	27
	24	Appeal from Default Judgement	27
	25	Unresolved Issues	27
	26	Examination of Evidence	27
	27	Comments on Evidence	28
	28	Request to Clarify by the Court	28
	29	Draft Judgement	28
	30	Objections to Draft Judgement	28
	31	Default Final Judgement	29
	32	Appeal from Default Final Judgement	29
	33	Final Judgement	29
5	Definitions of	of Truth and Falsehood	33
III	The Dif	ference	35
6	WCJ and IC	CJ Compared	37
7	Justice and	the Key People	43
IV	Thinkin	g Globally	47
8	National So	vereignty — Who Needs It?	49
9	False Moral	Equivalence	51
10	Tony Blair a	and the Rogue States	53
11	World Gove	rnment — Ideal or Painful Necessity?	55

CONTENTS v

V	Applying Principles to Cases	57
12	The Secret of Saddam's WMD and Establishing Trust in Government	59
13	Has the Fall of Baghdad Made the Iraq War Legitimate?	63
14	Clarification of the Legal Status of the US Presence in Iraq	67
15	Who Will Try Saddam Hussain, and Should it Matter?	71
16	The Trial of Saddam Hussain	75
VI	Interacting with Governments	79
17	Questions to World Leaders	81
18	G.W. Bush on The Middle East Conflict	85
19	President Bush's Speech on Resolving the Middle East Conflict	95
20	National Security Strategy of the United States of America 20.1 Introduction	113 114
	20.2 Overview of America's International Strategy	
	20.3 Champion Aspirations for Human Dignity	
	20.4 Alliances to Defeat Global Terrorism	
	20.5 Work with others to Defuse Regional Conflicts	
	20.6 Prevent Our Enemies from Threatening Us	
	20.7 Ignite a New Era of Global Economic Growth	
	20.8 Expand the Circle of Development	
	20.9 Develop Agendas for Cooperative Action	
21	Osama bin Laden's Message to the American People	253
Ind	dex	277

List of Tables

6.1	WCJ and ICJ Compared
7.1	Lord Denning's Decision in Birmingham Six
12.1	Justifying Iraq War by WMD 61
	Justifying Iraq War by Self-Defence 61
15.1	Counting Apples
15.2	Murder Case
17.1	Main Issues in the Middle East Conflict
18.1	G.W. Bush on The Middle East Conflict
19.1	President Bush's Speech on the Middle East Conflict 95
	Introduction to The National Security Strategy of the USA 114
20.2	Overview of America's International Strategy
20.3	Champion Aspirations for Human Dignity
20.4	Alliances to Defeat Global Terrorism
20.5	Work with others to Defuse Regional Conflicts
20.6	Prevent Our Enemies from Threatening Us
	Ignite a New Era of Global Economic Growth
	Expand the Circle of Development
	Develop Agendas for Cooperative Action
	OTransform America's National Security Institutions
21.1	Comments on Osama bin Laden Message 254

List of Figures

4.1	Stages of Civil Action in the World Court of Justice	30
4.2	Method of Proof in the World Court of Justice	31

Part I Why WCJ?

Chapter 1

Why the World Court of Justice?

Date of first publication: 2001–10–16

1.1 No Law between Nations

There was a time in human history, when disputes between private individuals were decided by use of force by the parties themselves. If one person disagreed with another he would kill, maim, beat or frighten his opponent into accepting his will. In most countries today this form of resolution of private disputes is not accepted, and the disputing parties are expected to submit their disputes to a court of law where such disputes are resolved following an established procedure.

Such form of resolution of disputes is restricted, however, to private individuals, and does not apply to disputes between established governments, or disputes between private individuals and groups and established governments.

So, if a dispute arises between established governments, a war erupts, and, if a dispute arises between people, who do not form an established government, and an established government, terrorism follows.

To stop the destruction of people and property by established governments (wars and matters of internal policy) and by non-governmental groups seeking to take the law in their own hands (terrorists) a supranational court of justice is required.

To be a genuine instrument of justice, rather than an instrument of politics of a country or group of countries, such court must be totally free from any political, ideological or any other influence and be totally and unconditionally objective and impartial (the rights of a single person are equal to the rights of the rest of mankind).

The operation of the court must consist solely of establishing facts on the basis of

evidence and evaluating them by applying strict mathematical logic to the fundamental principle: no person or group of persons (tribe, nation, organization, etc) have a right to interfere with the person, freedom or property of another person or group of persons without his consent, except for the purpose of prevention of such interference.

The need for a supranational body to resolve disputes between nations is generally recognized and attempts were made to establish such organizations.

1.2 Search for Supranational Justice

Following the Second World War, in an attempt to avoid wars, a group of states have established the Organization of the United Nations. But the Organization of the United Nations is not an independent body. Resolutions are based on votes, which express wishes of the majority — not on the principles of right and wrong, true and false.

The Super Powers (USA, Russia, etc.) have vetoes, and can block any resolution which is not convenient to them. This violates the main principle of justice — equality under the law.

The United Nations operate the International Court of Justice. The jurisdiction of that court is limited, however, only to resolution of disputes between states, members of the United Nations and Switzerland. The court has no jurisdiction to hear cases brought by non-governmental groups against established governments.

Moreover, although, formally, the judges composing the court are independent magistrates, and do not represent their governments, the technique of arriving at judicial decisions is based on judicial opinions and is by its nature subjective, i.e. reflects political philosophies, ideologies and personal prejudices of the judges.

There was also an attempt to set up a regional supranational court, the European Court of Human Rights. This court is restricted to Europe. Because the decisions of this court are based on judicial opinions, judgements of this court are not objective and impartial — they are influenced by personal views and prejudices of the judges.

It is clear that none of the above organizations is capable of being used as an instrument of justice necessary for prevention of wars and terrorism.

Nor can such court be established by politicians.

1.3 Politics and Justice

Definition 1.1 Politics is the use of the powers, given to politicians for the purpose of administration of their countries, to favor themselves and those on whose favours they depend.

Definition 1.2 *Justice is determination of rights between parties on the basis of total and unconditional objectivity and impartiality, strict truthfulness of facts, and strict logical validity of reasoning.*

Politics and justice are incompatible. Politics cannot be impartial and objective. Politics is favoritist and subjective by its very nature.

Truth and politics are incompatible. Politicians hide, distort and pervert facts to achieve their objectives.

Politics and logic are incompatible. Politicians pervert logic to arrive at conclusions favourable to themselves.

And it is only strictly impartial justice that can be respected at supranational level and be an effective means of resolving international disputes.

So, in a world inhabited by people whose judgements are affected by their emotions and prejudices and governed by politicians seeking to exploit these emotions and prejudices for their own ends, how can one set up a court, which will make wars, terrorism and politics things of the past?

1.4 Composition and Procedure

The court is an impersonal institution. There are no judges and no juries in the World Court of Justice.

The decisions of the court are based solely on the formal procedure defined by these rules. For that reason the decisions of the court do not depend on any person or group of persons or any personal views, wishes or opinions. Any human intervention is restricted solely to insuring that the procedure is followed correctly.

The court will be operated following a strict, totally open, mechanical procedure.

The objectivity and impartiality of the procedure will be insured by the structure of the procedure itself. All the proceedings will be in writing and publicly available to the whole world on the Internet.

Pleadings and evidence will be broken down into single-valued, clearly stated points of fact and items of evidence. All ambiguities will be clarified. All subjectivity and irrelevance will be removed. Each single-valued statement of fact will be strictly validated against the submitted evidence. The court will consider each validated fact separately and determine whether this fact singly or jointly with other validated facts constitutes an injustice.

1.5 Validation and Execution of Judgements

Because of the strictness and full public openness of the procedure, people all over the world will be able see for themselves the facts, the evidence and the logical reasoning used to arrive at the decision, and thus will validate the decisions of the court.

While the court has no possibility of enforcing its decisions (it has no armies and no police), its decisions will have intellectual and moral influence.

Honest and sincere people in national governments will welcome the World Court of Justice as an aid in their task of government, and will make use of the decisions and findings of the court in their work.

Bold page numbers indicate places where the meaning of words as used in this book is explained or defined.

```
1984
17th century
     Europe, 208
                                                second half, 93
9/11, 56, 105, 106, 119, 157, 164, 166,
          193, 201, 241, 243, 246, 247
                                            ability
     attacks, 230, 234
                                                human
       reasons for, 266
                                                   limitations of, 33
     bringing responsible to justice, 251
                                                to reason logically, 44
     deaths, 263
                                            abominations, 221
     failure to prevent, 247
                                            Abraham, 260
     hysteria, 234
                                            Abrahamic
     inspiration, 248
                                                religions, 163
     lessons of, 126
                                            absence
    not caused by poverty, 128
                                                 of peaceful means, 272
     paranoia, 183
                                            Absolute
     role of Afghanistan, 126
                                                 Authority
     shock of, 128
                                                   of the Lord, 268
     specter of, 201
                                            absolute
     trauma, 134, 138, 154, 183
                                                right, 135
                                            abuse
1948
                                                of judicial powers, 44
     expulsion of Palestinians, 86
                                                of people, 167
1967
                                                of power
                                                   crime of, 137
     Israeli occupation, 86
```

of powers, 218	justifications, 157
powers, 115	pattern, 190
abuses	unjustified, 246
of powers, 147	Afghani, 134
acceptance	Afghanistan, 49, 54, 76, 116, 126, 134,
of conditions, 178	137, 156, 157, 166, 167, 183,
access	192, 195, 246, 249, 271, 275
to means of destruction, 252	devastation, 117
Access to Justice, 10	devastation of
accountability, 132, 240	by US, 237
accountable, 102	did not attack US, 230
nation, 240	future of, 239
accumulation	liberated, 157
of successes, 155	Russian war, 168
accused, 73	war against, 230
act	Africa, 148, 158, 183, 184, 209, 214, 227,
adjudge to be void, 268	230, 233, 235
unlawful, 69	Growth and Opportunity Act, 209,
acting	211
alone, 160	size and diversity, 184
apart, 252	sub-Saharan, 211
action	African
against terrorism, 101	continent, 184
actions, 89	countries, 185
louder than words, 250	Development Fund, 222
adjusting	leaders, 143
and thriving, 252	states, 188
administration	Union, 185
of justice, 127	agendas
admittal	of cooperation, 241
of facts, 73	aggression, 125, 263
adult	against US, 246
supervision, 68	expansionist, 162
advantage	aggressive
perceived and real, 115	alliance, 141
adversity	designs, 190
turned into opportunity, 166	policies, 270
advertising	agreement
tools, 269	with Israel, 99
advice	within 3 years, 110
sound, 272	agreements
Afghan	between states
war, 51	redundancy, 210
justification, 167	to protect

US nationals from ICC, 250	world view, 229
agricultural	fighting together, 232
development	no need for if no enemies, 142
aid, 227	of terror, 118
sector, 213	allocation
agriculture, 209	of resources, 149
improved, 212	ambitions, 171
aid	expansionist, 191
measuring results of, 218	of Israel, 174
terrorists, 154	of national leaders, 245
AIDS, 129, 269	amenities
in Afghanistan, 130	of life, 268
in Saudi Arabia, 130	America, 99, 111, 144, 261
spread of, 163, 226, 227	land of freedom
Al-Aqsa	argument, 265
Intifada, 109	Latin, 148
al-Aqsa	North, 158
mosque, 263	ready to help, 111
al-Qaida, 116, 158	South, 158
expectations from US, 254	American, 134, see US
fighters, 157	Administration, 87
fighting US	administration, 56, 138, 150, 158
reasons for, 255	understands, 151
hunting down, 157	Blessing, 99
in Afghanistan, 192	bombings of Afghan cities, 157
issue of	citizens, 221
resolving, 158	civil aircraft, 120
sympathizer, 158	commitment, 173
alcohol, 161, 220, 227, 239, 250	concerns
alcoholism, 156, 269	about China, 240
Allah, 254, 266, 268	democracy
trust in, 272	not imposed, 123
Allah's	encouragement, 157
reward, 254	everything, 219
alliance	government, 164, 192
partners, 234	ideals, 250
alliances, 132, 134, 144, 154, 170, 195,	ideas
241	transplanting, 219
aggressive	interests, 184
criminal, 241	law, 139
defensive	occupation, 67
unnecessary, 241	people
allies, 138, 167, 176, 214, 244–246, 248	desire for peace and security, 275
and friends	friend of, 268

well-wisher of, 267	clutches of, 252
personnel, 272	anti-terrorism, 134
principles, 184	anti-terrorist
Servicemembers Protection Act, 250	paranoia, 202
status	anticipatory
in Iraq, 68	action, 201
steel	ANZUS, 234
industry, 213	appalling vista, 44
support, 104, 176	appeal, 63
for Israel, 192	Arab
system	countries, 192
defects of, 148	help, 105
system of government, 148	hostility
workers	cause, 83
protection from free trade, 213	cause of, 83
American people	leaders, 88
guilt	position, 90
per Osama bin Laden, 265	States, 99
americanization, 220	ties with Israel, 107
Americans, 49, 51, 56, 75	states
Americas, 230, 233	
	weakening, 220 volunteers, 157
anarchy	•
international, 54	world, 107
world, 55	Arab governments
Andean	surrender to Jews, 263
nations, 182	Arabian Gulf
Annan WF. 226	occupation of lands
Kofi, 226	paid by US taxes, 265
anthems, 86	Arabs, 260
anti-American	friendship with, 172
feelings	Arafat
reason for, 270	Yasser, 89, 133, 173, 176
sentiments, 116	removal of, 175
terrorism, 187	area
anti-ballistic	of governance, 266
defence system, 120	arguments, 63, 71
defence systems, 243	armed
anti-corruption, 91	forces, 256
anti-semitism	armies, 252
accusations of, 260	arms
anti-terror	control, 197
hysteria, 198, 251	Arms Race, 120
measures, 103	arrest, 64
paranoia, 184, 229	arrests

without reason, 271	evolution of, 148
arrogance, 195, 229, 274	authority, 101, 256
ascribing partners	legal, 61
to God, 267	avoidance
ASEAN, 235	of complications, 250
Asia, 158, 230, 233, 241	Axis
Asia-Pacific	of Evil, 189
Economic Cooperation Forum, 235	
region, 239	Babylon, 137
Asian	ancient, 163
and Pacific alliances	baddy, 63
obsolete and redundant, 235	Baghdad, 63
leaders, 143	balance
Muslim	of power, 116, 119, 126, 129, 136,
areas, 157	141, 144, 154, 164, 169, 186,
states, 188	191, 203, 220, 238
aspirations, 89, 240	favors freedom, 136, 228
assemble	that favors freedom, 245, 255
freely, 240	of terror, 186
assertions	ballistic
single valued, 60	missile, 202
assistance	technology, 186
economic, 167	missiles, 189, 194
for policy changes, 219	defenses against, 120
humanitarian, 167	ballot
objectives of, 218	box
political, 167	might of, 164
security, 167	banking
assuring	systems, 219
US allies and friends, 244	Barak
atrocities, 76	Ehud, 109
attack, 60	barons
attacking	pre-medieval English, 49
victim, 190	basic
attacks	education, 227
against Muslims	human
paid by US taxes, 265	values, 192
on Americans by al-Qaida	principles, 236
reasons for, 253	battles
auditing	winning, 252
independent, 102	beard, 202
Australia, 210, 230, 233, 234	long, 183
authoritarian	behavior
systems	patterns
Systems	patterns

modern, 163	paid by US taxes, 265
behavioral	blood
patterns, 220	value of, 134
Belgrade, 148	bloodshed, 115, 246
believing	more, 116
own propaganda, 167	boiling alive, 76
belligerence, 195	bombing
belligerent	Afghanistan
policies, 256	paid by US taxes, 265
rhetorics, 193	of mosques
benefit	by US, 271
of surprise, 247	residential areas
benefits	by Israel, 201
of labor, 114	wedding parties, 166
better	bombings
education, 223	suicide, 91
life, 213	Bonham's case, 9, 268
Bible, 8, 208	books
bilateral	religious, 8
agreements, 210	border
engagement, 184	controls
relationships, 238	improvement of, 166
with governments, 249	borders
biotechnology, 227	final, 105
bioterror, 166	secure and recognized, 108
Birmingham Six, 43	Brazil, 182
Bismark	Britain, 64, 111
Otto von, 230	British
black	historian, 105
hole, 186	in Palestine, 259
propaganda	media, 105
in Afghanistan, 161	Muslim youngsters, 134
blackmail, 186, 199	brutality, 188
Blair, 60	perverse, 93
rise to power, 60	brutalize
Tony, 53, 54, 59, 60, 81, 83, 143, 157	own people, 188
dossier, 194	building
spin, 162	future, 107
blame	police
Palestinians, 100	forces, 249
blind, 56	burglar, 63
justice, 170	killed, 63
blockade	burglary
of Iraq	attempted, 63, 64

bus	cause
driver, 124	just, 203
Bush, 60	of communist collapse, 166
deceives himself, 98	of violence, 82
George	primary, 93
enemies of, 202	causes
GW, 81, 83, 85, 88, 93, 103, 105,	and effects, 105
106, 189, 255	of conflict, 81, 89
answers, 85	of wars and terrorism, 167
legal identity of, 255	the three, 86
remarks, 85	causing
hates terrorists, 112	injustices
hope, 111	around the world, 251
irrational obsession, 112	centers
position, 90	of global power, 143, 241
speech	Central
June 2002, 95	America, 210
support for Israel, 97	Asia, 214
business	Cervantes
investment	Miguel, 183
encouragement, 205	chain of command
	unified, 104
call	challenge
to Islam	US to Israel, 179
of Americans by Osama bin	challenges, 144
Laden, 266	of 21st century, 168
callers	champion
to freedom, 270	aspirations, 141
Canada, 182, 212, 229	of persecuted, 275
capital, 105	championship
flows, 207	of human dignity, 150
markets, 207	changing
capitalists, 112	US foreign policy, 251
cardboard knives, 120	chaos, 53, 117
Caribbean	charity, 217
Basin, 211	cheap popularity, 174
Caspian	Chechen
region, 214	fighters, 157
cast	grievances, 152
rivalries, 143	Chechens, 275
casualties, 69	Chechnya, 54, 75, 137, 255, 275
civilian, 76	Russian atrocities, 261
catastrophic	war, 168
power, 186	chemical
power, rou	Chemical

104	
weapons, 194	civilized
children	nations, 155
adoption	clarification
by homosexuals, 268	of issues, 254
Chile, 182, 210	in disputes, 254
China, 60, 209, 216, 236, 240, 241	clarity
communist	of understanding, 82
legacy, 239	clash
democratic development, 239	inside a civilization, 250
strong	of civilizations, 250
US welcome, 239	class, 137
US trading partner, 240	society, 138
Chinese	structure
concerns	feudal, 50
about America, 240	struggle, 50, 143
leaders, 125	classes, 143
choice	clean
stark and simple, 112	coal, 215
to change	
government, 265	water, 223 clear
choose	
government, 114	priorities, 228
choose life, 112	climate
choosing	change, 215
our, 151	Clinton
Christianity, 163	President, 269
circle	cluster bombs, 76
of development, 217	coalition
circles	forces, 157
running in, 50	leadership, 228
civil	warfare, 232
society, 240	Coalition Forces, 51
wars, 249	coalitions, 197
local, 184	broad, 228
Civil Justice System	mission-based, 232
British, 10	of the willing, 133
civilian	of willing, 184
casualties, 201, 265	under
blame for, 265	NATO mandate, 232
population, 201	coercion
civilians	against US, 246
aggression against, 264	coffins, 272
civilization, 155 , 250	Coke
hallmarks of, 156	Lord, 9, 268
11411111411NO 01, 130	2010, 7, 200

Cold War, 119, 186, 187, 198, 199, 229, 233, 235, 236	compensation, 16, 45, 83, 102 monetary, 83
concepts and assumptions, 186	competent, 115
era, 187	competitions, 211
armies, 245	completeness, 33
rhetorics, 114	complex
collapse	negotiations, 249
of empires, 244	comprehensive
collateral	peace agreement, 111
damage, 201, 264	plan, 166
Colombia, 182	compromised
colonies, 230	by terror, 98, 118
combating	concepts
terror, 118	need to define, 149
comfort, 217	of past centuries, 214
command-and-control	conclusion, 275
economies, 206	conclusions, 73
comment	concrete
invited to, 82	benchmarks, 223
comments	conditions
on NSS report 2002, 113	unacceptable, 190
committed	confidence, 89, 91, 211
to peace, 106	conflict
common	between law and politics, 8
action, 233, 241	cause
dangers, 121	removing, 82
interests, 235	causes of, 81, 86
law, 9, 268	in War on Terror, 256
principles	Israeli-Palestinian, 51
supremacy of, 241	mediation, 185
reason, 268	on terms of others, 151
responsibility, 185	resolution, 272
right, 268	resolved, 82
strategic	conflicts, 217, 245, 272
interests, 238	causes, 45
swearing, 127	international
values, 122, 235	current resolution, 275
communism	over resources, 185
collapse of, 161, 166	reason for, 229
unworkable, 141	regional, 142
Communist	resolution of, 198
Party	resolved
Chinese, 240	by justice, 142
compassion, 217	resolved by
compassion, 217	10301 vod by

instina 172	controlictory 266
justice, 172	contradictory, 266
taking sides in, 272	control
with Muslim minorities, 164	of US
confront	by Jews, 268
corruption, 104	convenient, 56, 65
confrontation, 186	cooperation, 150, 186, 195
confronting	defense
failure, 218	and diplomatic, 235
congress, 210	coordinating
conquer, i	intelligence with US allies, 247
the world	Coral Sea
failures, 244	Battle, 235
conquest	core
of other nations, 136	development
consequence	assistance, 220
management, 198	issues
consequences	that divide, 108
failure to foresee, 130	corrupt
of actions, 202	leaders
Conservative Party, 143	support of, 272
conservatives, 60	corrupted
constant	youth, 161
threat, 244	corrupting
constitution, 102, 268	Muslim nations, 250
constitutional	corruption, 123, 176, 188
framework, 102	fighting, 220
constructive	of Muslim lands, 263
dialogue, 275	cosmetic change, 100
consultation, 241	council
of partners, 228	estates, 143
consumer	counterproliferation, 196
products, 269	counterterrorism, 186, 239
contract, 9, 15	cooperation, 234
breach of, 15	counting
inviolability of, 143	apples, 72
contract-based	court
commercial	appeal, 63
interactions, 218	British, 63
contracts	composition of, 71
in writing, 208	decisions
contradicting	enforcement, 5
pure nature, 268	validated, 5
contradiction	of law, 3
in terms, 174	operation of, 3
III (CIIII), 1/7	operation of, 3

procedure	improvement, 227
openness of, 5	crossroads
system, 91	of radicalism and technology, 119,
systems, 249	186
courts	cruel uncle, 68
independent, 8	Crusade
need for, 55	Bush began, 275
of law	Crusades, 183, 275
absence, 103	crutches, 56
US, 139	Cuba, 134
cover up, 44, 51	Cuban
Creator, 268	missile
credibility	crisis, 199
loss of, 162	cumulative
credit	effect, 159
ratings, 207	current
crime, 16, 64, 123	injustices, 261
organized, 140, 183	customs, 221
paying price for, 259	cut into little pieces, 76
prevention, 52	
crimes, 51, 64, 259	damage, 83, 101
against humanity, 256	by primitive equipment, 243
as injustice, 16	danger
by governments, 54-56, 65, 145,	to freedom, 186
153	dangerous
by nation states, 53, 65	human interference
right to commit, 55	with climate, 214
criminal, 51	world, 189
act, 67, 258	dangers
case, 63	extraordinary, 211
conspiracy, 141, 241	gather, 202
courts, 271	Dark Forces, 246
gangs, 117	dark shadow, 96
criminality	Day of Judgement, 267
of governments, 56	dead, 56
criminals	burglar, 64
become terrorists, 183	shot, 63
crises	death, 64
financial, 207	and fear, 95
management, 171	desiring more than life, 273
critical	imminent, 64
medicines, 211	deaths
crop	of Iraqi children, 263
yields	of millions, 112
J. 10.100	

11	174 176
decades	statement, 174, 176
of destruction and bloodshed, 113 of peace, 139	demands
* ·	concessions to, 153 democracies, 209
deception, 59 decision	,
	democracy, 8, 9, 100, 111, 114, 122 , 149,
in Birmingham Six, 44	227, 270
decisions	American book of, 270
judicial, 11 decisive	demagogical use of word, 238 embracing, 176
	O .
victory, 114	European style, 50
Declaration of Independence, 147 dedicated	obsolete, 143
effort, 105	practicing, 98 slogan of, 122
*	transplanting, 143
deep anger and despair, 111 default	working, 102
failure to remedy, 272	democratic
defeat	elections, 148
admittal of, 195	government, 111
defeating	hemisphere, 182
adversary, 244	institutions
terrorism, 167	development of, 150
defence, 186	meaning of, 173
of victims of injustice, 263	nations, 232
defend	political
freedom, 247	systems, 185
defenders	demonstrations, 153
of liberty, 148	denials, 73
defending	Denning
oppressed, 267	Lord, 43
our nation, 117	deny
persecuted, 267	sanctuary, 118
defenses	Department
beyond challenge, 243	of Defense, 166
defensive	transforming, 246
alliance, 141	of Homeland Security, 166
define	of State, 249
non-politically, 114	dependence
definition	on judges, 72
of responsibilities, 175	on leaders, 229
deflation	despicable
in Japan, 206	state, 267
delivering	despotism, 8
results, 223	destroying
demagogical	economy, 266

1	1
destruction, 69, 115, 186, 246, 259	dictators, 75
and bloodshed, 200, 252, 253	dictatorships, 188
for decades to come, 251	difference
in Palestine	between politics and crime, 140
paid by US taxes, 265	differences
more, 116	ethnic, 50
of Afghan villages	ethnic and religious, 185
by US, 271	national, 50
of civilians, 83	different
of houses, 53	approach, 81
of nature, 270	diffusing
of Palestinian property, 179	military
of people, 82	confrontation, 181
of people and property, 3	dignity, 111
of property, 82	human, 141
of towns and villages, 266	affronts to, 170
destructive, 55	diplomacy, 165, 197
force	diplomatic
use of, 270	institutions, 250
policies	Director
of US, 270	of Central Intelligence, 248
powers, 187	disarmament
deter	by all nations, 125, 244
threats against US, 244	disarming
deterrence, 137, 186, 199, 200, 203, 245	of nation states, 171
failability, 247	disaster, 82
failure of, 244	discarding
devastation, 259	of false ideas, 267
in Palestine, 101	discrediting
developed	ideology, 151
world, 102	disease, 183
resources, 226	diseases
developing	spread of, 269
countries, 222, 227	disgrace
world, 206	fleeing in, 275
development, 226, 227	dishonest, 55, 99
aid, 204, 218	dishonesty, 34 , 153
assistance, 223	in business, 269
failure, 218	dispute, 81
grants, 224	•
projects	disputes
support, 224	arising among humans, 7
dictating	between governments, 3
few against many, 88	between individuals, 3

between individuals and govern- ments, 3 between nation states, 228 escalation of explosive, 170	dominate, i domination, 199 double standards, 271 drug
human	addiction, 156
reasons for, 254	addicts, 129
international, 55	cartels
private	violence, 182
resolution of, 3	trafficking, 182
regional	drug-addiction, 269
active engagement in, 170	drugs, 161, 220, 227
resolution	demand for, 182
by peaceful means, 254	in US, 269
resolution of, 7	in wealthy nations, 128
resolving	supply of, 182
by violence, 152 in courts of law, 152	duel, 127 , 155
within family, 7	dumping, 212
disqualification, 62	dynamism
distinct	of economy, 252
roles, 266	
distinction	East Turkistan, 137, 239
between domestic and foreign af-	economic
fairs, 251	activities, 206
divide-and-rule	bankruptcy, 275
policies, 187	development, 176, 208, 223
Divine	freedom, 205, 238
books	encouragement, 220
mutability of, 267	growth, 204–206, 213, 214, 218, 225
divorce, 15	institutions, 100
DMA, 182	openness, 125
Doha	policies, 205, 220
US	policies and challenges, 205
agenda, 209	reform, 204
doing	stability, 204
something about it, 121	stagnation, 102
dollars	economies
per donor, 218	of scale, 232
domestic	Economy, 206
capital, 225	economy
governance, 249	thriving, 111
industries	educate
help to adjust, 213	children, 114

education, 99, 204, 205, 220, 222, 224,	to violence, 106
227, 249	encourage
for people's benefit, 266	change, 148
egoistic	encouragement
demands, 174	of enmity, 247
Egypt, 99, 105, 137	encouraging
elected, 115	terrorism, 103, 106
election	endangering
campaigns, 270	life, 272
electoral	ending
systems, 249	the conflict, 180
eliminate	endless
terrorism, 104	war, 158
eliminating	and terror, 168
causes, 81	enemies, 56, 75, 76, 117, 119, 134, 138,
of terrorism, 104	141, 203, 229, 236, 244
elimination	of civilization, 118, 202
by removal of causes, 151, 165	of the people, 134, 202
elusive	seeking WMD, 186
targets, 247	enemy
embarrassment, 62	existence of, 186
embittered	fire, 166
few, 138	energy
embitterment	security, 214
sources of, 139	enforceable
embrace	agreements, 212
democracy, 104	enforcement
emergency	of judgements, 20
management	enjoining good, 267
systems, 166	enmity
state of, 62	and wars, 236
emerging	enthusiastic, 56
markets, 207	entrepreneurial
threats, 202	activity, 205
emotional	energy, 252
	9.
reactions, 81	entrepreneurship
emotions, 5	enablement, 220
empire, 242	environment 212
empires	protection, 213
20th century, 119	envy, 218
collapse of, 242	envy-based
replaced empires, 258	philosophy, 217
employment, 218	epidemics, 226
encitement	equal

-:4:	and aslanias 220
citizens	and colonies, 230
of one world, 133	central and eastern, 148
justice, 145	not all world, 230
under the law, 51, 52, 182, 221	part of world, 233
equaling	united against the rest, 233
power	western, 9
of US, 247	without
equality	borders, 230
before the law, 106, 168	European
of all people, 267	allies, 184, 185, 233
of nations, 229, 236, 239, 241	colonies, 230
under the law, 243, 251	Court of Human Rights, 4
under the law, 9, 15, 121, 148, 164,	Empires, 187, 188
171, 172	Imperialism, 187
equivalence	partners, 206
moral	structural
false, 51	barriers, 206
espionage, 165	Union, 49, 56
establishing	dream of, 230
justice, 267	world view, 230
establishment, 45	even-handedly, 65
establishmentarianism, 270	even-handedness, 272
Ethiopia, 185	events
ethnic	current, 81
ambitions, 175	evidence, 5, 72, 73
conflicts, 76	evaluation, 73
purity	evil
ideal of, 138	empires, 274
violence	rid the world of, 151
in India, 138	evil deeds, 64
ethnic-religious	exchange
rivalries, 143	rate
ethnicity, 255	policy, 207
EU, 102, 212, 230	executive-congressional
creature of two wars, 233	partnership, 210
foreign policy	expectations, 89
identity, 233	expel
Euro-Atlantic	from houses, 87
community, 236	experience
Russian integration in, 237	painful, 50, 56
eurocentric	experimenting, 232
world view, 233	explanation, 254
Europe, 158, 206, 229, 235, 241	exploit, 82
against whom, 230	exploit, 82 exploitation
agamst whom, 250	exploitation

of female sexuality, 269	fanatic
export	extremists, 256
controls, 197	farm
opportunities, 240	exports, 212
expulsion, 99, 259, 265	farmer, 63
of Arabs and Christians, 175	fast track system, 10
of Arabs by Jews	Fatah, 92
justification of, 258	fate
extermination, 99	of Mankind, 275
of hated groups, 112	of the world, 105
	favoring
facts	friends, 75
distorted, 161	favoritism, 8, 116, 149
establishing, 3	FBI, 166
fabricated, 161	fear, 52
false, 82	as motivator, 120
primary, 73	of punishment, 51
proved, 56, 73	Federal Government, 133
true, 82	feelings
validated, 5	of injustice, 82
failed	feudal lords, 8
policies, 218	feudalism, 143
strategy	fiction, 183
indicators, 218	fierce
failing	in anger, 151
states, 138	fight
failure	against invasion, 69
to give explanation, 254	against terrorists, 104
to justify actions, 254	corruption, 204
to redress wrongs, 254	for just peace, 136
fair	for lawful rights, 105
competition, 212	in Cause of Allah, 254
trading	prepared to, 274
practices, 213	terror, 97, 100
fallacious	terrorism, 167
fabrication, 260	fighting
false, 9, 34	a group of people, 151
moral	between tribes and nations, 258
equivalence, 51, 52	by al-Qaida
falsehood, 33	reasons, 254
family	terror, 131
as unit of society, 7	terrorism, 238
of nations	terrorists and tyrants, 116, 256
world view, 229	final
WOIIG VICW, 22)	IIIIMI

cottlement 105	investment 225
settlement, 105	investment, 225
final status agreement, 110	sympathizers, 69 forestall
finance, 103	
financial	hostile acts, 201
crises, 207	fornication, 267
management	foundations of government, 11
of DoD, 246	Founding Fathers, 148
markets	France, 50, 56
uncertainty, 207	Franklin
sector	Benjamin, 268
policy, 207	freaks
systems, 205	and cranks, 117
fire, 56	free
fiscal	and open societies, 136
and monetary policy, 207	enterprise, 114
flags, 86	flow of commerce, 238
fleets, 252	flow of ideas, 165
flexibility	market
of command structures, 232	economics, 206, 208
focusing	world-wide, 209, 210
efforts and resources, 165	markets, 142, 204, 225
foe	society
of civilization, 158	value of, 136
foes	speech, 145
of humanity, 133	trade, 125, 142, 204, 205, 208, 209,
football	225
hooligan, 105	area of Americas, 209
forbidding	benefits of, 213
evil, 267	decline of, 208
force	in Europe and Americas, 208
configurations, 232	promotion of, 209
use of, 3	free-market
forces	democracy
of freedom, 114	in Russia, 237
of terror, 96	
, , , ,	freedom, 15, 49, 111, 113, 114, 140, 144,
transformation and modernization, 232	204, 217, 237, 239, 270
_	advancing, 149
forcing	and fear at war, 168
inhabitants	as slogan, 173
from houses, 258	benefits of, 126
foreign	economic, 114, 125
affairs, 271	fighters, 105
aid, 149	for some
assistance, 149	at expense of others, 245

infringed, 147	of the war, 251
infringements of, 173	future
limitations on, 16	will tell, 115, 117
meaning of, 16	wiii teli, 113, 117
non-negotiable, 133	G-7, 206
of movement, 179, 218, 219	action plan, 207
of religion and conscience, 150	G-8
of the strong	agreement, 197
to oppress the week, 168	gambling, 267, 269
of worship, 145	gangland, 132
peace and security, 242	gangsters
political, 114	power-drunk, 117
political and economic, 185	quarrel, 117
promoting, 149	gases, 270
to make a living, 208	gathering
freedom-loving	dangers, 202
people, 114	gay
freer	vote, 143
and juster, 147	Gaza, 86, 111
trade, 213	general
friendly	approach, 85
fire, 201	disarmament, 240
triends 56 116 134 138 141 154 167	general hallof 115
friends, 56, 116, 134, 138, 141, 154, 167, 176, 188, 203, 244–246, 252	general ballot, 115 generosity, 217
176, 188, 203, 244–246, 252	generosity, 217
176, 188, 203, 244–246, 252 allies	generosity, 217 genocidal
176, 188, 203, 244–246, 252 allies enemies, 186	generosity, 217 genocidal war, 138
176, 188, 203, 244–246, 252 allies enemies, 186 and enemies, 119, 132	generosity, 217 genocidal war, 138 genocide, 75, 162, 192
176, 188, 203, 244–246, 252 allies enemies, 186 and enemies, 119, 132 of Satan, 254	generosity, 217 genocidal war, 138 genocide, 75, 162, 192 in Chechnya, 163
176, 188, 203, 244–246, 252 allies enemies, 186 and enemies, 119, 132 of Satan, 254 US, 270	generosity, 217 genocidal war, 138 genocide, 75, 162, 192 in Chechnya, 163 in Palestine, 163
176, 188, 203, 244–246, 252 allies enemies, 186 and enemies, 119, 132 of Satan, 254 US, 270 friends and allies	generosity, 217 genocidal war, 138 genocide, 75, 162, 192 in Chechnya, 163 in Palestine, 163 in Palestine and Chechnya, 168
176, 188, 203, 244–246, 252 allies enemies, 186 and enemies, 119, 132 of Satan, 254 US, 270 friends and allies mentality, 252	generosity, 217 genocidal war, 138 genocide, 75, 162, 192 in Chechnya, 163 in Palestine, 163 in Palestine and Chechnya, 168 of Palestinian people, 162
176, 188, 203, 244–246, 252 allies enemies, 186 and enemies, 119, 132 of Satan, 254 US, 270 friends and allies mentality, 252 friendships, 170	generosity, 217 genocidal war, 138 genocide, 75, 162, 192 in Chechnya, 163 in Palestine, 163 in Palestine and Chechnya, 168 of Palestinian people, 162 of smaller nations, 137
176, 188, 203, 244–246, 252 allies enemies, 186 and enemies, 119, 132 of Satan, 254 US, 270 friends and allies mentality, 252 friendships, 170 frivolity, 8	generosity, 217 genocidal war, 138 genocide, 75, 162, 192 in Chechnya, 163 in Palestine, 163 in Palestine and Chechnya, 168 of Palestinian people, 162 of smaller nations, 137 geo-political
176, 188, 203, 244–246, 252 allies enemies, 186 and enemies, 119, 132 of Satan, 254 US, 270 friends and allies mentality, 252 friendships, 170 frivolity, 8 of politicians, 268	generosity, 217 genocidal war, 138 genocide, 75, 162, 192 in Chechnya, 163 in Palestine, 163 in Palestine and Chechnya, 168 of Palestinian people, 162 of smaller nations, 137 geo-political experiments, 118
allies enemies, 186 and enemies, 119, 132 of Satan, 254 US, 270 friends and allies mentality, 252 friendships, 170 frivolity, 8 of politicians, 268 frivolous	generosity, 217 genocidal war, 138 genocide, 75, 162, 192 in Chechnya, 163 in Palestine, 163 in Palestine and Chechnya, 168 of Palestinian people, 162 of smaller nations, 137 geo-political experiments, 118 German, 193
allies enemies, 186 and enemies, 119, 132 of Satan, 254 US, 270 friends and allies mentality, 252 friendships, 170 frivolity, 8 of politicians, 268 frivolous super-power, 251	generosity, 217 genocidal war, 138 genocide, 75, 162, 192 in Chechnya, 163 in Palestine, 163 in Palestine and Chechnya, 168 of Palestinian people, 162 of smaller nations, 137 geo-political experiments, 118 German, 193 Germany, 56, 101, 174
allies enemies, 186 and enemies, 119, 132 of Satan, 254 US, 270 friends and allies mentality, 252 friendships, 170 frivolity, 8 of politicians, 268 frivolous super-power, 251 frivolously	generosity, 217 genocidal war, 138 genocide, 75, 162, 192 in Chechnya, 163 in Palestine, 163 in Palestine and Chechnya, 168 of Palestinian people, 162 of smaller nations, 137 geo-political experiments, 118 German, 193 Germany, 56, 101, 174 global
allies enemies, 186 and enemies, 119, 132 of Satan, 254 US, 270 friends and allies mentality, 252 friendships, 170 frivolity, 8 of politicians, 268 frivolous super-power, 251 frivolously attacking nations, 116	generosity, 217 genocidal war, 138 genocide, 75, 162, 192 in Chechnya, 163 in Palestine, 163 in Palestine and Chechnya, 168 of Palestinian people, 162 of smaller nations, 137 geo-political experiments, 118 German, 193 Germany, 56, 101, 174 global affairs, 235
allies enemies, 186 and enemies, 119, 132 of Satan, 254 US, 270 friends and allies mentality, 252 friendships, 170 frivolity, 8 of politicians, 268 frivolous super-power, 251 frivolously attacking nations, 116 fuel	generosity, 217 genocidal war, 138 genocide, 75, 162, 192 in Chechnya, 163 in Palestine, 163 in Palestine and Chechnya, 168 of Palestinian people, 162 of smaller nations, 137 geo-political experiments, 118 German, 193 Germany, 56, 101, 174 global affairs, 235 climate, 214
allies enemies, 186 and enemies, 119, 132 of Satan, 254 US, 270 friends and allies mentality, 252 friendships, 170 frivolity, 8 of politicians, 268 frivolous super-power, 251 frivolously attacking nations, 116 fuel economy, 215	generosity, 217 genocidal war, 138 genocide, 75, 162, 192 in Chechnya, 163 in Palestine, 163 in Palestine and Chechnya, 168 of Palestinian people, 162 of smaller nations, 137 geo-political experiments, 118 German, 193 Germany, 56, 101, 174 global affairs, 235 climate, 214 consensus, 236
allies enemies, 186 and enemies, 119, 132 of Satan, 254 US, 270 friends and allies mentality, 252 friendships, 170 frivolity, 8 of politicians, 268 frivolous super-power, 251 frivolously attacking nations, 116 fuel economy, 215 fundamental	generosity, 217 genocidal war, 138 genocide, 75, 162, 192 in Chechnya, 163 in Palestine, 163 in Palestine and Chechnya, 168 of Palestinian people, 162 of smaller nations, 137 geo-political experiments, 118 German, 193 Germany, 56, 101, 174 global affairs, 235 climate, 214 consensus, 236 disarmament, 251
allies enemies, 186 and enemies, 119, 132 of Satan, 254 US, 270 friends and allies mentality, 252 friendships, 170 frivolity, 8 of politicians, 268 frivolous super-power, 251 frivolously attacking nations, 116 fuel economy, 215 fundamental interests and values, 252	generosity, 217 genocidal war, 138 genocide, 75, 162, 192 in Chechnya, 163 in Palestine, 163 in Palestine and Chechnya, 168 of Palestinian people, 162 of smaller nations, 137 geo-political experiments, 118 German, 193 Germany, 56, 101, 174 global affairs, 235 climate, 214 consensus, 236 disarmament, 251 economic
allies enemies, 186 and enemies, 119, 132 of Satan, 254 US, 270 friends and allies mentality, 252 friendships, 170 frivolity, 8 of politicians, 268 frivolous super-power, 251 frivolously attacking nations, 116 fuel economy, 215 fundamental	generosity, 217 genocidal war, 138 genocide, 75, 162, 192 in Chechnya, 163 in Palestine, 163 in Palestine and Chechnya, 168 of Palestinian people, 162 of smaller nations, 137 geo-political experiments, 118 German, 193 Germany, 56, 101, 174 global affairs, 235 climate, 214 consensus, 236 disarmament, 251

energy, 214	governance
enterprise, 118	by name calling, 54
fund	within nation states, 114
for HIV/AIDS, 226	government, 77, 114
government	above law, 53
principles, i	abuse of powers, 115
initiative, 209	activities of, 76
policeman, 64	aid, 225
power	British, 65
centers of, 241	by demagogues, 123
problems, i	by name calling, 53
security, 121, 206	by organized crime, 124
commitments, 250	by truth, honesty and justice, 56, 131
solutions, i	by violence and politics, 56
tasks, 138	changes of
terror	periodic, 148
	communications, 60
combating, 183 terrorism, 141, 158	competent and honest, 60
	decisions
as general phenomenon, 165	
meaning, 191	making, 60
terrorist organizations	duties, 16, 71
disrupting and destroying, 158	duty of, 217
thinking, i, 138	elections, 122
union, 185	evil, 64
view, 238	federal, 117
village, 191	gangland model, 132
war on terrorism	global, 49
US-Russian cooperation, 237	principles, i
world, 49	honest, 45, 116
order, 158	ensuring, 123
view, 134	honest and competent, 52
Global Terrorism, 246	honest competent and effective, 100,
globalized	148, 174, 204, 220, 266
world, 167, 251	ideology, 118
God, 174	informers, 118
Goebbels, 162	institutions, 50
Golan Heights, 86	instruments of, 147, 149
Goldenberg	Israel friendly, 98
Susan, 91	ministers, 102
good	moderate and modern, 163
governance, 185	non-political, 148
intention, 139–141	oppressive, 63, 67
good outcome, 69	powers, 16, 75
goody, 63	abuse, 76

abuse of, 125	Greater
abused, 124	Israel, 161, 198
limiting, 124	Zionists, 191
procurement, 209	
	greatness
reorganization, 166	national, 125
repressive, 65	source of, 239
respectable, 162	greenhouse
responses and responsibilities, 205	gas
responsibilities of, 16	concentration, 214
subsidies, 212	emissions, 214
supranational, 53, 55, 56, 228	grievances, 139, 164
system of	legitimate
abuse of, 148	refusal to redress, 156
trust in, 60	grow
trusting, 62	in prosperity, 129
US, 65	growth, 211, 226
without politics, 49	Guantanamo, 272
governments	Guardian, 91
aggression	guided
defence against, 153	missiles, 243
become terrorists, 183	guilty
established, 117	of murder, 73
hereditary, 124	party, 263
honest competent and effective, 185	Gulf
inadequate and corrupt, 204	war, 194
national, 56, 228	gullibility
removal of, 263	•
ruling	of people, 250
	Hamas, 92, 99, 106
justly, 220	Haram al-Sharif
granting	
rights, 267	Sharon visit, 109
grants	harbor
results-based, 224	terrorists, 154
great	legitimate, 154
nation	hardship
lie, 267	alleviation of, 217
power	harm, 186
competition, 236	serious, 53
powers, 121	suffering, 53
and terrorism, 121	harmful
compete, 228	acts, 71
compete in peace, 121	hasbara, 161
concept, 236	hatred
struggles, 114	atmosphere of, 193

for US	homeland defences, 118
reasons for, 250	homicide bombings, 106
of few, 96	homosexual
haughtiness, 274	family, 268
haven	homosexuality, 129, 156, 161, 163, 220,
for terrorists, 167	239, 267
head	decriminalization of, 129
of state, 256	homosexuals, 143
health, 204, 205, 222, 224	honest, 55, 115
care, 220, 223	enterprise, 102
of US, 182	governance, 226
regulations, 212	government, 102
hereditary, 115	trade, 142
Hezbollah, see Hizbullah	honesty, 34 , 146
high-risk	as instrument of government, 146
groups, 129	the best policy, 162
highjackers, 126	with oneself, 272
historical	honor, 267
right	hope, 89, 91, 182
to Palestine, 260	hopes
significance, 261	have power, 112
history	of many, 96
distance of, 151	horizon, 93
human, i, 243	hostile
not kind, 146	activities
of Mankind, 75, 244, 268	towards states, 165
responsibility to, 151	propaganda, 165
will judge harshly, 120	states, 198, 247
Hitler, 146, 162, 202, 229	hostilities, 254
Adolf, 112	against Americans, 261
elected	hostility
democratically, 123	· · · · · · · · · · · · · · · · · · ·
popularity, 174	ensuring constant supply of, 251 towards Arabs, 193
Hitlerism, 200	towards Muslims, 193
	Human
HIV, 129	
HIV/AIDS, 211, 224, 226, 227	Rights, 271
problem	statistics, 271
of US and China, 239 spread of, 239	human actions
•	outcome of, 113
Hizbullah, 99, 106	· · · · · · · · · · · · · · · · · · ·
homeland	results of, 115 dignity, 140, 145, 149, 183
defense, 246	
protection, 251	affronts to, 170
security, 165, 196	cause of, 150

avistance 171	ideals 146
existence, 171	ideals, 146
freedom, 116	identification, 255
history, 242, 258	of people with governments
hopes, 111	in wars, 265
ingenuity, 120	ideological
no limit to, 247	downfall, 275
no limits to, 248	fashions, 9
judgement, 148	ideologies, 161, 164
life, 206	20th century, 242
nature	ideology, 164
deficiencies of, 174	political, 51
needs, 224	IDS, 59, 60
progress, 206	ILO, 213
rights, 114, 220, 240	IMF, 102, 207
denial by governments, 150	imminent
suffering, 172	danger, 59
Human History, 76	threat, 200
humanitarian	concept, 200
assistance, 102, 176, 220	immoral
catastrophes, 249	acts, 267, 269
cause of, 249	immorality, 162
relief	acts of, 269
requirements, 249	immunity
humiliation, 262	for US, 271
humility, 228, 229	impartiality, 5, 45, 150, 172, 272
hunger, 227	imperial
hypocrisy, 149, 153, 239, 272	grandeur
of politicians, 106	delusions, 199
hypocrites, 272	past, 236
hypocritical, 99	spirit, 236
hysteria, 244	imposing
anti-terrorist, 119	own values, 184
41117 (01101101, 11)	will
ICC, 250	on US, 247
ICJ	impossible
compared with WCJ, 37	to be performed, 268
IDA, 222–224	imprisonment
ideal, 55	without trial, 271
glorious, 56	imprisonments, 270
idealism, 56	improving
idealisms, 146	lives, 222
idealist dream, 55	impunity, 55
idealistic	inability
view, 56	to deter, 198
v10w, 50	10 detel, 170

inaction	of government by people, 264
not an option, 202	information
incentives, 205	collection
incest, 269	methods, 248
incomes, 204	gathering, 247
incompatible, 266	openness, 240
politics and justice, 5	operations, 246
politics and logic, 5	technology, 227
politics and truth, 5	infrastructure
incompetent, 99	of democracy, 143
independence, 102	ingenuity
independent	human
judiciary, 103	no limit to, 243
states, 230	inheritance
India, 60, 164, 216, 236, 241, 261	rules of, 208
and Pakistan disputes, 181	injury
as world power, 238	imminent, 64
Indian	injustice, 5, 21 , 106, 217 , 270
economic reforms, 238	as motivator, 120
nuclear and missile programs, 238	by those in power, 164
Ocean, 238	feelings of, 82
oppression	prevention of, 228
in Kashmir, 260	redress, 82
individual	redress of, 69
human	injustices
rights, 240	alleged by Osama bin Laden, 255
responsibility, 125	by governments, 154
Indonesia, 181	by national governments, 245
industrial	current
espionage, 212	redress of, 258
goods, 209	existing
growth, 208	failure to redress, 251
pollution, 270	future
relations, 221	prevention of, 258
waste, 270	redress of, 117, 228, 254, 255
industrialized	suffered by Palestinians in 1948, 257
countries, 206	innocent, 51
infallibility	innovation, 205, 246
of the US government, 165	inquiries, 62
of US, 250	insecurity, 244, 245
infectious	reason for, 229
diseases, 166	institutions, 89, 91
influence	instruments
economic and political, 115	of government, 114

insurgency, 53	US, 140
integrated	of others, 228
intelligence, 202	interference
integrating, 232	in methods of education, 272
intelligence with	with person, 15
defense, 247	with property, 17
integration, 182	interlocking
integrity, 75	strategies, 184
of legal system, 44	International
of police, 44	community, 176
intellectual	Court of Justice, 4
property	Criminal Court, 103
rules, 211	Law
intelligence, 118, 186	respect of by US, 271
and law enforcement	international
fusion of information, 248	aid, 132
capabilities, 248	community, 87, 103, 212
community, 247	support, 160
infrastructure, 184	cooperation, 149, 249
initiatives, 248	disputes
US use of, 247	resolution of, 256
warning	institutions, 149, 171, 185, 249
framework, 248	law, 68
warning and analysis, 248	obligations, 133
intended action	politics, 249
statement of, 272	relationships, 168, 171
intent	scene, 11
to kill, 64	terrorism, 161
intention	trade, 212
of killing, 64	treaties
intentions	violations, 189
and results, 219	tribunal, 75
declared and real, 115	internationalism
inter-continent	American, 140
rivalries and hostilities, 233	Internet, i, 5, 253
inter-state	intoxicants, 267
rivalry, 199	in US, 269
interaction	inventing
with nations, 249	laws, 268
interest, 267	investment, 209, 211
interests, 244	inwardly
common	digest, 140
advancement, 232	Iran, 75, 117, 168, 189, 194
national, 76, 171	Iranian, 106

revolution, 168	extending, 87
Iraq, 53, 76, 106, 117, 189, 192, 196, 255,	creation of, 259
271	defends herself, 97
and Iran, 194	democracy, 107
invasion, 67	destruction of, 106
liberation, 67	expansion, 220
war, 51, 59, 63–65	expansion of, 161
justified, 59	financing by US, 117
pattern, 190	taxes, 265
pretext, 190	friendship with, 172
support for, 59	Greater, 97, 99, 105
WMD, 59	identity, 107
Iraqi	identity and democracy, 178
government, 192	peace and security, 173
people, 75	right to defend itself, 255
regime, 51	security, 107
irrational	settlements
obsession	activities, 107
with capturing or killing, 251	State of
irrelevance	creation, 257
removal of, 5	destruction of, 83
irrelevant statements, 33	preservation of, 83
Islam, 163, 267	ties with Arab States, 107
call to	withdraw to pre-September 28, 2000
by Osama bin Laden, 266	positions, 107
encourages terrorism	Israeli, 134
idea that, 164	armed
Osama's view, 256	forces, 87
Islamic	Army, 104, 179
Jihad, 92, 106	freedom of, 173
Law, 254	attacks, 87
law, 130	attacks on Palestinians, 86
Nation, 274	disregard, 179
fight with, 272	expansionist
party in Algeria, 270	plans, 180
states	incursion, 83
weakening, 220	incursions, 92
Ummah, 260	Lobby, 100
world, 111	military operations, 101
Islamist militants, 92	military rule, 101
isms, 51	•
131113, 21	occupation, 133, 192
	occupation, 133, 192 1967, 108
Israel, 60, 85, 189	1967, 108
	-

to destroy al-Aqsa, 263	jobs, 204
position, 90	joint
•	operations, 246
rejected	<u>*</u>
Saudi Plan, 107	Jordan, 87, 99, 105, 210
settlements, 179	Judaism, 163
settlers, 87	Judea and Samarea, 175
war, 175	judge
war against	incapacity
Palestinians, 116	due to involvement, 172
war against Palestinians	judgement
US encouragement of, 237	clouded, 128
withdrawal, 83, 175	perverse, 44
Zionists, 111	judges, 8
Israeli-Palestinian	British, 10
conflict, 172	duties of, 72
Israelis, 51, 81	need for, 55
live in terror, 96	personalities, 71
victimized by terrorists, 97	judicial
issue	capacity, 217
of fact, 72	opinions, 4
of law, 72	process, 71
issues	system, 254
in disputes, 254	judiciary, 249
, -	independent, 176
Jangi fort, 271	jurisdiction
Japan, 206, 234, 235, 271	declarative, 20
Japanese	just, 55, 67, 217
banking	cause, 154
system, 206	solution, 83
Jenin, 133	justice, 4 , 55, 102, 147 , 164, 254
Jerusalem, 108, 260	act of, 153
East, 87, 175	blind, 45, 52, 170
Jesus, 146, 260, 272	brought to, 64
Jewish	confusions about, 217
aggression	criminal, 71
in Lebanon, 260	denied, 147
Jews, 112	enforcement, 52
historical connection with Palestine,	
258	formalization of, 8 fundamental principle, 51
historical right of	going back in history, 258
<u> </u>	• •
to Palestine, 258	informal, 7
support of, 272	instrument of government, 135
Jihad, 256, 263	issues of, 217
in the way of Allah, 267	not favor, 135

perverted, 147	king, 49
principles, 56	will of, 8
reliable, 103	kings, 8
supranational, 55, 65, 169	British, 50
system, 55	knights
systems	and crusaders, 183
national, 55	knowledge
tribal, 8	of laws, 268
justification, 67	spread of, 211
of violence	Korea, 148, 234
by US, 179	Korean
justified	peninsula, 239
allegations, 275	Kurds, 75
justify, 51	Kyoto
justifying	agreement, 270
Iraq war	1.1
by self-defence, 61	labor
by WMD, 61	and environmental concerns, 213
war, 193	disputes, 221
justly	lack
failure to act, 217	of actions, 89
	language, 34
Kabul, 156	of manners and principles, 266
Kashmir, 275	understanding, 34
hostilities, 261	US understands, 266
Kenya, 105, 185	Latin America, 182, 188
Khoumeini, 168	law
kill, 75	above will of kings, 8
all terrorists, 105	abstract, 8
killed	American, 139
or imprisoned, 158	and order, 55, 142
killers	international, 55
trained and determined, 108	world, 67
killing, 259, 265	between nations, 3
civilians, 266	codes of, 8
deliberate, 71, 72	common, 9
of prisoners	enforcement, 64, 118, 184, 249
by US, 271	equal under, 50, 134
Palestinians, 179	global
progress, 96	rule of, 50
unjustified, 72	international, 67
killings, 53, 75	regard for, 189
kindness	none above, 140
to others, 267	of human behaviour, 8
<i>'</i>	

of warfare, 201	legitimacy
rule of, 50, 65	of government attacks on civilians,
Shari'a, 130	264
single, 49	legitimate
source of, 60	acts, 75
supranational, 53, 54, 56, 75, 117,	aspirations, 89
132, 141, 144	grievances
supremacy of, 9	addressing within a political pro-
taking in own hands, 3	cess, 152
lawful	prevent peace, 152
authority, 69	targets, 261, 262
combatants, 264	lending
inhabitants, 258	strategy, 207
lawless	less
city, 117	rich, 218
lawlessness, i	lethal
laws, 268	environment, 184
civil and criminal, 8	letter, 81
in force, 72	before action, 254, 272
leaders	liberate, 65
national, 56	mankind, 115
personalities of, 229	liberated
League of Nations, 11	meaning of, 157
learning, 111	liberating
Lebanon, 108, 271	Iraq, 67
Israeli occupation of, 261	Liberation
left	Struggle, 98, 103
and right, 182	liberty, 98, 114, 184
legal	can blossom, 111
codes, 249	habits of, 204
document, 275	political and economic, 116
framework	possibilities to expand, 149
deficiency, 207	licence
workable, 219	to commit crimes, 53
reform, 204	lie, 34
representative	lies, 262
-	life
of country, 256 of terrorist group, 256	and death, 112
status, 67	of hope, 111
system, 117, 204	of Mankind, 253
legality	stranger than fiction, 183
importance of, 69	light
legislation	of hope, 97
social, 9	Likud, 107

minimalist chicatives 175	forming 140
minimalist objectives, 175	favoring, 140
Olami, 198	rule, 143
ideologues, 200	vote, 147
philosophy, 180	will of, 148
position, 90	make
Zionists, 161, 190, 220	world
limiting	better, 140
powers of national governments,	malaria, 211, 226
171	malnutrition, 227
limits	maneuver
on state power, 145	and expeditionary forces, 246
list	Manila
of facts, 254	government, 272
literacy	Mankind, i
and learning, 227	interests of, 275
living	security of, 229
and prospering in peace, 146	manners, 267
on 2 USD a day, 217	manslaughter, 63, 64
with fear and funerals, 110	manufacturing, 209
loans	market
non-performing, 206	access, 211
local	economics, 101
government	economies, 206
institutions, 249	incentives, 206
infrastructure, 226	market-access
officials, 102	negotiations, 209
logic	markets
application of, 4	closed, 204
long-distance	emerging, 206
deployment, 245	open, 204
looting	marriage, 15
in Afghanistan, 157	heterosexual, 130
love	martial law, 103
	martyrdom
of God, 267	in death, 200
lucky, 218	Marxist
modness 200	
madness, 200	thinking, 206
maintain	view
US forces	of economics, 206
in Asia, 235	mass-casualty
Major	dangers, 166
John, 60	master-and-servant, 218
major powers, 116	mature
majority	and responsible way, 234

Mau-Mau, 105	defeat, 275
maximize	dictatorships, 117, 123
injury, 247	force, 244
means	forces
of defense	transformation of, 202
lead to new weapons, 243	logistical
peaceful, 82	support, 234
measurable	might, 81, 118, 136, 164, 244
goals, 223	reliance on, 243
results, 223	necessity, 83, 103
measure, 208	operations, 177
measuring	options
success, 223	range of, 246
mercy, 267	power, 113, 118
method	presence, 83
political, 81	strength, 242
Mexican, 134	unparalleled, 115
Mexico, 182, 212	strikes
Middle East, 158, 161, 190, 194	retaliative, 82
conference, 82	style training camps, 126
Conflict, 45, 157	technology, 190
primary cause of, 257	Millennium Challenge Account, 129, 220
conflict, 81, 95, 170	mindset
causes, 81	of US government, 186, 193
failure to resolve, 172	tribal, i
might	mini skirts, 161, 220
is right, 11, 121, 147, 164, 168, 176,	minimal force, 64
243	minority
militant	favoring, 140
approach, 89	will of, 148
organisations, 89	miracle
military	of Qur'an, 267
action, 67, 271	miscarriage of justice, 45
adventures, 252	miseducated, 264
aggression, 125, 199	miseducation, 207
build-up, 247	misery
capabilities	perpetuation of, 218
pursuit of by China, 239	misinformed, 264
pursuit of by US, 239	missile
command, 166	defense, 186
competition	mistakes, 62
dissuading, 244	Mitchell
stopping, 244	Committee, 179
coup, 115	recommendations, 107
coup, 115	recommendations, 107

D 100	C
Report, 108	from one country to another, 258
mob rule	Mowlam
organized, 143	Mo, 59
modern	Muhammad, 260
life, 251	Prophet, 267
society, 252	Mujahideen, 271, 275
modernity	multi-national
test of, 163	entities, 49
monarchies, 208	multi-party
money	elections, 102
flow of, 106	multilateral, 211
for development, 225	development
moral	banks, 223, 224
advice, 266	or supranational, 132
education, 227	multinational
equivalence	fighting forces, 232
false, 51	murder, 63, 64, 71, 72
guide, 266	case, 73
imperative, 217	murderers
influence, 5	rhetorical use of word, 127
obligation, 223	Muslim, 254
opinion, 52	conquest of Palestine, 260
person, 275	countries, 111, 163
principle, 208	governments of, 256
superiority, 60	modernizing, 163
values	country
abandonment of, 239	bombing, 134
morality, 8, 111, 162	Nation
rejection of, 161, 220	per Osama bin Laden, 255
morally	world, 163
inferior, 51	future, 250
superior, 51	youth, 161
Morocco, 210	westernizing, 220
Moscow	Muslims, 255, 256, 260, 275
Treaty on Strategic Reductions, 236	representation of, 256
Moseley	mutual
Oswald, 123	assured
Moses, 260	destruction, 186
motivating	interest, 272
acts of terror, 247	respect, 195
motivation, 247	myth
motivations	Socialist, 50
of enemies, 187	mythical
moving	meanings, 143

•	
myths	core mission, 231
of right left and center, 220	creature of Cold War, 233
mamaatias	membership
narcotics	expansion, 232
trafficking, 182	military forces, 232
Nation	mobile
of Honor and Respect, 273	forces, 231
of Martyrdom, 273	redundant, 229
of Monotheism, 272	rewards
nation, 137	of success, 233
single, 56	self-defence
state, 67	clause, 230
states	structures
rival, 136	and capabilities, 231
nation-state	NATO-Russia Council, 237
rise of, 228	natural obligations, 15
National	necessity
Security	unpleasant, 55
Council, 166	need
national	to specify, 261
and homeland security, 248	needs, 240
frontiers, 218	of today's world, 252
governments, 192, 228	Netanyahu, 270, 271
greatness	Benjamin, 162, 183, 194, 255
pursuit of by China, 239	books and speeches, 268
interests, 75, 119, 134, 221, 230,	influence over US policy, 264
233, 236–238, 242, 249, 251	NCLCI speech, 188
selfish, 138	theory, 256, 264
leaders, 75, 76	blueprint, 251
police, 132	doctrine, 191, 252
policies, 219	theories, 190
reform	new
challenge of, 220	condition
security, 204	of life, 252
spending on, 251	era, 113
US, 201	leaders
success, 114	emerge, 110
nationalism, 138	path, 97
nations	technologies, 227
alone, 132	New Labour, 143, 161
equal	New York, 272
under the law, 125	New Zealand, 234
must act, 106	newspapers, 59
NATO, 133, 229, 233, 235	Nigeria, 185

nightmares	world view, 148
of 20th century, 238	objectivity, 5, 45, 150, 172, 272
no peace and security	obligation
for US and the world, 244	contractual, 217
noble cause	obligations
dying for, 164	of US, 136
non-democracies, 231	obsessive
non-government	irrational behavior, 119
groups, 228	obstacles, 184
non-governmental	to free and secure world, 252
organizations, 167, 249	occupation, 100, 265, 272
para-military groups, 191	American, 49
non-intervention, 53, 261	end to
non-political	distant, 111
governments, 185	financing, 99
non-super-powers, 119	permanent, 107
non-violent	offense
means, 104	as best defense, 165
nonprofits, 224	official
nonproliferation	corruption, 102
commitments	one world, 134, 185
of China, 240	one-party
North	rule, 240
Korea, 60, 189, 194, 196, 235	ongoing
Koreas, 194	disputes, 212
Northeast	•
	open market, 101
Asia, 245	markets, 181
Northern Alliance, 156	societies, 211
NSS Report, 253	
September 2002, 113	society, 111
comments, 114	opening
nuclear	societies, 143
and missile capabilities, 238	to commerce, 225
arsenals, 186	openness
club, 190	in China, 240
power, 215	of procedure, 5
weapons	Operation
acquisition, 194	Enduring
nutrition, 224	Freedom, 234
0.10.100.100	operational
OAS, 133, 182	challenges, 245
oath	demands, 232
of allegiance, 106	opinion
objective	moral, 52

public, 171	own
opinions, 5, 11	property, 114
judicial, 8	
opportunities, 144	PA
opportunity, 112, 182, 211, 217	destruction of, 97
for US, 136, 139	position, 90
hoarded, 204	packing
oppression, 69, 76, 259, 263, 270	luggage, 272
of small nations	painful
by super-powers, 247	experience, 141
resisting, 168	Pakistan, 60
oppressive	and India, 181
acts, 53	democracy, 123
government, 67	in War on Terror, 181
governments, 69	Palestine, 53, 54, 85, 137, 246, 249, 255,
organized	275
violence, 254	blood poring out of, 260
orphans, 260	democratic, 107
Osama	handing over to Jews, 263
bin Laden, i, 157, 183, 201, 254	handover by British, 259
case, 255	historical right to
extradition, 230	of Jews, 260
justification of seeking redress,	independent and democratic, 173
259	provisional
legal identity of, 255	state, 104
message to Americans, 253	regaining, 263
responsibility for 9/11, 230	under military occupation, 259
bin Laden's claims	Palestinian, 82, 134
against Arab States, 262	Arabs, 258
Osama's	Authority, 87, 88, 175
expectations from US, 266	composition, 88
our boys, 60	structure, 88
outcome	buildings
of human actions, 113	destroyed, 87
of trial, 71	civilians
outdated path	killed, 87
of China, 239	economy
of US, 239	to develop, 107
outright conflict, 116	government
outstanding	loyal to Israel, 174
disputes, 202	structure and composition, 174
Oval Office, 269	grievances, 152
Owen	hostilities, 82
David, 59	injustices

at the hands of Israelis, 259	pandemics, 226
institution-building, 108	pandering
land	to envy, 218
Israeli control over, 259	panic
leadership, 97	reaction, 246
legislature, 101	panicking
<u> </u>	· •
organizations, 83	public, 121
parliament, 102	paranoia, 198, 244
prisoners, 177	anti-terrorist, 119, 138
property	Quixotic, 183
destruction of, 177	US, 200
reform, 108	paranoias
refugees, 107, 177	as motivators, 119
plight and future, 108	pariahs, 196
resistance, 97, 173, 177, 192	Parliament
revenues	act of, 9, 268
release of, 107	parliaments, 8
rights, 111	parties
security services, 104	identities, 255
State, 99	partnerships, 195
state, 100, 107, 176	party
recognized, 175	politics, 50
viable, 107	Pashtun
viable and credible, 179	village, 183
vision, 172, 259	path
vision of, 108	of action, 121
struggle, 192	of progress, 241
suffering, 102	of reform, 104
territories	open
occupation of, 83	for all, 141
terrorism, 83	straight, 50
villages	Patriot Act, 118
destruction of, 257	patriotic
Palestinians, 51, 81, 133, 275	fervor, 60
cry of, 260	pattern
defend themselves, 97	of US aggression, 190
expulsion of	patterns
from houses, 257	of behavior, 163
in 1948, 83, 86	peace, 45, 76, 113, 144, 184, 237, 239
grow miserable, 97	and freedom, 173
live in squalor, 96	eventual, 93
paltry	favors liberty, 136
price	in the world, 168
selling at, 262	living in, 85
=	=

operations, 185	personalities
plan	of judges, 73
Saudi, 83	persons
process, 106	and nations, 218
derail, 86	perverse judgements, 10
illusion, 101	pesticides, 227
setbacks, 108	PFLP, 92
side of, 106	Philippines, 234
stable, 117	piety, 267
through equality and justice, 199	pillage
treaties, 110	of treasures, 263
world, 65	pillar
peaceful	of economics, 208
and secure	pipe
world, 248	dream, 176
means, 81, 89, 104, 228, 253	piracy, 162
relations, 140	plan
world	workable, 93
blueprint, i	planning
people	contingencies, 246
destruction of, 83	pleadings, 5
in power, 130	plenty, 217
voices and interests, 174	plot
will of, 8	of Satan, 254
permanent	plotting, 236
occupation, 178	police, 55
permission	policies
to fight, 254	legal and regulatory, 205
to revenge, 266	policing
perpetual	of governments, 221
insecurity, 136	political
persecuter, 266	breakup, 275
person, 217	catch-phrase, 97, 104
evil, 63	concepts, 114
honest, 63	considerations, 81
immoral and dishonest, 51	demagogue, 121
legal meaning, 17	demagogues, 147
moral and honest, 51	aggressive, 122
personal	demagogy, 173
freedom, 269	ends, 82
freedoms, 240	establishment
gain	western, 163
of rulers, 188	
	freedom, 238, 239
weapons, 132	ideology, 51

influences, 148	development, 207
institutions, 100	poor
interference	countries, 224, 226
with economic activities, 208	pop
lexicon, 163	music, 161, 220, 250
methods, 81	popularity, 220
objectives, 165	porous
justifying, 164	borders, 184
openness, 240	positions, 90
paranoia, 118	pre-September 28, 2000, 179
parties, 270	potential
process, 93	aggressors, 232
reasons, 65	danger, 165
restrictions	for investigations, 250
on trade, 208	poverty, 183, 206, 211
slogan, 175	and terrorism, 126
slogans, 173	reduction, 218, 219
theorists, 220	relative, 218
political convenience, 9	Powell
politicians, 5, 8, 59	Colin, 93, 108
boasting, 220	power
gay, 220	as duty, 124
never admit crimes, 153	as privilege, 124
powers of, 55	of the people, 143
politics, 4 , 11, 45, 52, 54, 60, 75–77, 112,	over creatures and land, 268
114 , 126, 129, 140 , 144, 149,	positions of, 75, 228
164, 204, 207	vested in supranational government,
abandonment of, 131	228
abolished, 143	powers
abolition of, 148, 149	great, 116
alternative to, 60	limitation of, 49
crime of, 17, 76	limited, 49
elimination of, 115	to limit, 56
eradication of, 165, 205	pre-1967
incompatible with	borders, 107
dignity, 140	precision, 33, 275
freedom, 140	strike
peace, 140	capabilities, 246
left-right, 50	preempting, 160
of past centuries, 187	preemption
of power, 8	as pretext
party, 50	for aggression, 202
politics-free .	legitimacy of, 200
economic	preemptive

actions, 201	pro-American, 65
attack, 200	pro-Israeli, 65
strike	problems
doctrine, 201	of today, 187
strikes, 203	regional and global, 235
prejudice, 45, 148	procedure
prejudices, 5, 149, 171	inadequacies of, 148
of judges, 20	proceeding
preservation	deliberately, 202
of institutions, 242	productive
preserve	investments, 224
peace, 116	potential
President	unleashing, 219
of USA, 255	productivity
pressure	growth, 222
for reform, 218	progress
groups, 9, 81, 153	make, 107
preventing	path to, 140
terrorism, 166	towards security, 179
prevention, 226	Prohibition, 269
and response options, 247	promiscuity, 239
better than cure, 130	promiscuous
programs, 226	sex, 129, 250
pride, 195	promise
prime principles, 10, 64, 242	and opportunity, 183
principle	promote
fundamental, 4	image, 60
principles, 267, 272	proof
and values	of guilt, 73
defence of, 270	propaganda, 162
must be defined, 149	age of, 161
of global government, i	black, 161
of law, 72, 73	to improve US image, 250
of liberty, 136	propagandists, 162
priorities, 217	property, 217
prison	defending, 63
of fear and subdual, 262	destruction of, 83
prisoners	private
of ideologies, 238	sacredness of, 143
private	rights, 173, 208, 211
individuals, 228	Prophets, 260
sector, 207, 222, 224	of Allah, 260
privileged	prophets, 8
few, 204	prosecution

of HS by ICC 250	Our'on 9 209
of US by ICC, 250	Qur'an, 8, 208
prosperity, 111, 114, 182, 184, 204, 206	22:39, 254
protecting	2:169–171, 274
America, 167	3:139, 273
Americans, 250	4:76, 254
protests, 153, 264	58:21, 274
proved	61:9, 274
guilty, 75	63:8, 273
provisional	9:13, 273
state, 104	immutability of, 267
provocation, 63	
provoking	race, 137
hatred, 247	racism, 138
pseudo-classes, 143	in Europe, 138
psychiatrist, 200	in the US, 138
public	radical
and private sector, 166	organisations, 89
diplomacy, 249	Radio Sawa, 161
health, 211, 249	raising
crisis, 226	living standards, 222
information	raping
efforts, 250	in Afghanistan, 157
office, 62	rates
scrutiny, 68, 82	of growth, 218
punish, 64	reactive
crimes, 75	posture, 198
punishing	real
criminals	goal, 191
to protect people, 266	power, 102
enemies, 56	realism, 56
punishment, 72, 265	of US, 171
purity, 267	Reality
purpose	Force of, i
of inquiries, 33	reality, 34, 34
put on trial, 76	reason
Putin, 75, 124	for belief, 82
1 4411, 73, 121	reasonable, 55
questionnaire, 81	reasons
WCJ, 85	conflict
quiet	continues, 96
argument, 241	for al-Qaida's fight, 254
Quixote	rebellious
de la Mancha	smaller nations, 137
Don, 183	rebuilding
Doil, 100	recuirding

Afghanistan, 167	initiatives, 209
recognition	integration, 211
of states, 86	organizations, 167
recommendations, 179	peace and stability, 235
reconciliation, 76	reject
recruitment	terror, 104, 176
and retention	relations
in DoD, 246	among great powers, 136
redress, 69	good, 116
injustice, 82	relationship
of injustices	between people, 217
by peaceful means, 275	relationships
redressing	between nation states, 114
grievances, 139	religion
injustices, 245, 266	of all prophets, 267
committed by governments, 152	of Unification of God, 267
wrongs	religions, 8
means of, 254	religious
reform, 100	belief, 255
true, 100	incantation, 106
reformed	remarks
Palestinian	Bush
government, 176	GW, 85
reforming	remote
states, 185	sensing, 246
refugees	removing
Palestinian, 257	governments, 67
refusal	
Terusar	renewable
to submit, 168	
	renewable
to submit, 168	renewable energy, 215
to submit, 168 regime	renewable energy, 215 representative
to submit, 168 regime changes, 155, 165, 190, 246	renewable energy, 215 representative government, 238
to submit, 168 regime changes, 155, 165, 190, 246 regime change concept, 270	renewable energy, 215 representative government, 238 representatives
to submit, 168 regime changes, 155, 165, 190, 246 regime change	renewable energy, 215 representative government, 238 representatives of the parties, 256
to submit, 168 regime changes, 155, 165, 190, 246 regime change concept, 270 regimes, 106, 188	renewable energy, 215 representative government, 238 representatives of the parties, 256 repression, 53
to submit, 168 regime changes, 155, 165, 190, 246 regime change concept, 270 regimes, 106, 188 changed, 53	renewable energy, 215 representative government, 238 representatives of the parties, 256 repression, 53 of Muslim minorities, 240
to submit, 168 regime changes, 155, 165, 190, 246 regime change concept, 270 regimes, 106, 188 changed, 53 harbor	renewable energy, 215 representative government, 238 representatives of the parties, 256 repression, 53 of Muslim minorities, 240 repressions, 75
to submit, 168 regime changes, 155, 165, 190, 246 regime change concept, 270 regimes, 106, 188 changed, 53 harbor terrorism, 155	renewable energy, 215 representative government, 238 representatives of the parties, 256 repression, 53 of Muslim minorities, 240 repressive
to submit, 168 regime changes, 155, 165, 190, 246 regime change concept, 270 regimes, 106, 188 changed, 53 harbor terrorism, 155 non-modern, 161	renewable energy, 215 representative government, 238 representatives of the parties, 256 repression, 53 of Muslim minorities, 240 repressive governments, 150
to submit, 168 regime changes, 155, 165, 190, 246 regime change concept, 270 regimes, 106, 188 changed, 53 harbor terrorism, 155 non-modern, 161 region	renewable energy, 215 representative government, 238 representatives of the parties, 256 repression, 53 of Muslim minorities, 240 repressive governments, 150 repugnant, 268
to submit, 168 regime changes, 155, 165, 190, 246 regime change concept, 270 regimes, 106, 188 changed, 53 harbor terrorism, 155 non-modern, 161 region hopeful, 88	renewable energy, 215 representative government, 238 representatives of the parties, 256 repression, 53 of Muslim minorities, 240 repressive governments, 150 repugnant, 268 requests
to submit, 168 regime changes, 155, 165, 190, 246 regime change concept, 270 regimes, 106, 188 changed, 53 harbor terrorism, 155 non-modern, 161 region hopeful, 88 secure, 88	renewable energy, 215 representative government, 238 representatives of the parties, 256 repression, 53 of Muslim minorities, 240 repressive governments, 150 repugnant, 268 requests US
to submit, 168 regime changes, 155, 165, 190, 246 regime change concept, 270 regimes, 106, 188 changed, 53 harbor terrorism, 155 non-modern, 161 region hopeful, 88 secure, 88 regional, 211	renewable energy, 215 representative government, 238 representatives of the parties, 256 repression, 53 of Muslim minorities, 240 repressions, 75 repressive governments, 150 repugnant, 268 requests US disregard of by Israel, 179

resilience	rewarding
of institutions, 252	friends, 56
resistance, 263	Reza Pahlevi, 117
resolution	rich
of disputes	and poor, 217
obstacle to, 249	right, 9
of international disputes, 275	and wrong
resolutions, 179	<u> </u>
	language of, 145
resolving	sense of, 8, 10
conflict, 81	left and center, 161
issues	of governments
through justice, 180	to wage wars, 264
resources	of oppressed to return aggression,
allocation of, 211	263
of US	to declare wars, 59
finite, 171	to defend, 69
respect	interests, 59
for life, freedom and property, 192	oneself, 87, 131
for property	to defend oneself, 106
private, 145	to expel Arabs from Palestine, 258
for women, 145	to Palestine, 260
respectable	to resist, 69
government, 163	to security, 110
respecting	to strike back, 266
values judgements and interests, 252	right-center-left, 143
responsibilities, 99	righteousness, 267
avoidance of, 130	rights
magnitude of, 170	of smaller nations, 103
of individuals, 130	rise up to the occasion, 134
of nations, 130	risk
of the parties, 175	of inaction, 201
of US, 134, 136	risk-averse
to protect, 53	adversary, 199
responsibility, 105	rival
diminished, 63	continental
state of, 128	unions, 230
of adults of unsound mind, 17	warlords, 117
of minors, 17	rivalries
of people for government, 264	among major powers, 170
responsible, 55	roadmap
for their own actions, 50	Middle East, 45
restitution, 16	rogue
retaliation, 266	regimes, 194, 196
revenge, 76, 260, 263	state, 53
	•

states, 53, 187–189, 195, 197–200,	trial, 71, 75
244, 246	safe
theory, 271	from terror, 217
role	world, 116
crucial, 180	safer
Romans, 260	place, 63, 203
ruined cities, 56	world, 217
rule	sanitation, 224
of law, 91, 103, 111, 117, 129, 145,	Saracen, 183
168, 176, 181, 205, 211, 212,	Satanic Invention, 269
220, 222, 241	satellites, 188
in China, 240	satisfaction
the world, 65	of requirements, 242
rules	Saudi
and principles, 140	Peace Plan, 83, 90, 93, 97, 99, 104,
of procedure, 139	108
rumors, 161	scandals, 62, 123
Russia, 9, 56, 60, 164, 209, 236, 241	scheming, 236
and US	school caretaker, 76
strategic interests, 236	science, 227, 246
democratic future, 122	regulations, 212
US	secure
relationship, 236	and peaceful world, 236
war against Chechnya, 125	security, 45, 105, 113, 144, 182, 237, 239
Russian	challenges, 245
atrocities	environment, 186, 187
in Chechnya, 260	of US, 182
distrust of US, 237	world, 65
Federation, 75	seeking to redress injustice, 259
imperial	seizures
politics, 187	of property, 53
oligarch, 124	self
war	help, 171
against Chechnya, 237	know your own, 146
weaknesses, 236	self-deception, 34, 167
Russians, 157	self-defence, 63, 64, 67, 162, 230, 266
in Afghanistan, 168	act of, 64
in Chechnya, 272	issue of, 230
in Checiniya, 272	in Osama bin Laden's case, 255
sabotage, 165	meaning of, 64
Saddam Hussain, 63, 65, 67, 69, 75–77,	non-violent options, 162
117	right of, 97, 263
capture, 76	use of violence in, 255
crimes, 75	self-glorification, 75
CHIIICS, 13	sch-giornication, 13

salf protection 100	shields
self-protection, 190	
Semites, 260	and spears, 243
sense of right and wrong, 68	shooting
separating	at wedding parties, 166
religion	shotgun, 63
from policies, 268	sieve
separation of issues, 266	with bottom removed, 118
September 11, <i>see</i> 9/11	signature
September 28, 2000, 107	by US
serious	value of, 272
about peace, 112	simplistic
services, 209	picture, 114
settlement	sincere
just and comprehensive, 175	friend, 268
settlements, 177	person, 275
extension, 87	statement, 176
Jewish, 83	sincerity, 267
sex	Singapore, 210, 234
abuse of, 269	single
orgies, 163	nation, 242
tourism	sustainable
American and European, 226	model, 114
trade, 269	skills
sexual	of people, 252
immorality	slavery, 162
in US, 269	sleeves
liberation, 129	empty, 56
promiscuity, 129, 156, 161, 163, 220	slogans, 147
sexually	to justify powers, 114
transmitted	sober
diseases, 163, 227	analysis, 241
shadowy	sobering
networks, 117	experience, 56
Shari'a, 268	soccer
law, 130	hooligan, 105
making supreme law, 263	social
prevention of, 262	acceptability, 9
Sharon	rivalries, 143
Ariel, 85, 87, 107, 268	sectors, 224
man of peace, 266	socialism, 9, 143, 161, 208, 217
position, 90	international, 146
Sharon's	national, 146
visit to al-Aqsa, 263	unworkable, 9
Shia, 75	socialisms, 51

socialist	powers of, 49
economies, 206	responsibilities
experiment, 206	of states, 160
ideal, 9	sovereignty, 49, 53, 105, 182
ideology, 138	loss of, 49
myths, 138	national, 49, 55
thinking, 9	parliamentary, 9
socially	personal, 55
acceptable, 140, 163	principle of, 53
societies	unipolar, 49
free and open, 117	Soviet
sodomy, 163, 250	bloc, 247
soldiers	imperial
American, 54	politics, 187
British, 54	Security, 166
Israeli, 54	system, 118
Russian, 54	collapsed, 118
solidarity, 150	Union, 111, 118, 141, 187, 188, 229
Solomon	collapse of, 186
Kingdom of, 258	states of former, 237
solution, 81, 82	Soviets, 120, 187
workable, 81	fate of, 275
Somalia, 271	Spain, 183
US attack on, 260	speak
US involvement in 1992-95, 261	freely, 114
sound	Specter
advice, 272	of Global Terrorism, 186, 231, 244
sources	spectrum
of power, 116, 228	of threats, 248
South	speculations, 71
Africa, 185	spin, 161
American	doctors, 162
dictatorships, 117	spirit
Asia, 181	of humility, 228
Korea, 235	sponsoring
Lebanon, 86	terror, 200
Southern	terrorism, 191
Africa, 210	sponsors
Philippines, 272	of global terrorism, 165
sovereign, 49	squadrons, 252
kingdoms, 133	squader squader
monarch, 50	national
national	resources, 188
power, 137	
power, 137	stability

in emerging	Ummah's wealth, 262
markets, 207	stealth deceit and murder, 151
restoration of, 170	steps
stable, 217	security, 91
Asia, 238	towards Palestinian statehood, 104
Stalin, 146, 202, 229	stolen property
Joseph, 112	return of, 154
worship of, 174	stop
standards	terror, 175
of justice, 103	violence, 86, 93
starving	stopping
Muslims of Iraq, 263	debauchery, 267
State	immorality, 267
Department, 249	lies, 267
state	oppression, 267
failure, 53	straight path, 50, 141
frontiers, 118	strategic
of the world, 114	friendships imply enmities, 236
security	thinking, 236
Soviet, 118	strategies, 238
terrorism, 192	Straw
statehood, 86, 100	Jack, 82
statelessness, 200	strengthen
statements	alliances, 141
false, 34	strengthening
irrelevant, 62	defenses
making, 34	to win war, 251
misinterpreting, 34	strike
misunderstanding, 34	first, 198
of evidence in, 61	strikes, 221
of fact, 60	struggle
of law, 60	long and painful, 50
of relevance, 60	non-violent, 149
validity of, 62	of ideas, 250
states	over ideas, 137
weak, 126	stupidity, 229
stating	sub-regional
case	organizations, 185
in writing, 254	sub-Saharan
specifically, 252	Africa, 185
status	subdual, 272
legal, 67	subjective
quo, 100	factors, 171
stealing	subjectivity, 45
=	•

submission	level, 219
to Gods's Laws, 267	police, 65, 125, 131, 142, 153, 221
success, 113, 254	232, 233, 235, 236, 243, 244
suffering, 117, 217	surpassing
alleviation of, 170	power
human	of US, 247
minimization of, 170	survival, 68
Summit	symbol
of the Americas, 182	of American
super powers, 4, 103, 111	prosperity, 242
super-power, 134	symbols
arrogance, 184	of US commitments, 245
club, 238	sympathy
dominated	public, 63
world, 238	Syria, 87, 106, 108
friends and allies, 133	fights terror, 108
megalomania, 252	ngines terror, 100
might of, 243	Taiwan, 148, 209
philosophy, 252	Relations Act, 240
single, 203	taking sides, 81
super-powers, 119, 164, 165, 188, 189,	Talib, 158
197, 203, 237	Taliban, 49, 51, 134, 156, 167, 246
ganged up together, 137	crime, 157
superiority	government, 130, 156, 157, 192
conventional	hunting down, 157
of US, 199	war
support	against, 158
charges of, 261	targeting
supranational, 11, see government	of innocents, 200
= =	
court, 20, 65	of attacks, 201
government, 53, 133, 184	
institutions, 229	tax
institution, i	rates, 205
institutions, 55, 125, 171, 202, 229,	the rich, 174
233, 235, 236, 241, 248	taxation, 17
justice, 153, 155, 221	taxes
need for, 272	people's control over spending, 264
justice and police, 240	taxing the rich, 17
law, 69, 142, 149, 185, 243	teacher
law and police, 251	training, 227
legal	technological
framework, 207	opportunities, 232
system, 232, 236	technology, 246
legal system, 243	regulations, 212

4.11 1 102	
teddy bears, 103	not inevitable, 251
Temple	not opposing, 104
Mount, 175	Palestinian
Sharon visit, 109	cause of, 83
Third, 175	phenomenon, 183
temporary	prevention, 166
access, 245	prevention of, 245
temptation, 51, 76	putting end to, 201
terror, 77	reasons for, 245
condemning, 176	redundancy of, 265
fight off, 88	redundant, 104
justified by no cause, 152	removing causes of, 167, 243
terrorism, 52, 55, 115, 117 , 134, 139,	sanctuary, 160
164, 173, 207, 228, 254	sponsorship, 160
acts of, 164	support, 160
and injustice, 127	use of
and war	to achieve political goals, 155
difference between, 152, 162	vulnerability to, 251
and wars	will disappear, 104
justified by self-defence, 152	terrorist
as enemy, 151	and extremist
as phenomenon of nature, 151	groups, 182
by national governments, 155	attack, 234
campaign against, 168	attacks
caused by governments, 265	fueling of, 247
defeat of	cells, 158
by military action, 238	danger
disappearance of, 199	intelligence against, 248
eliminating, 131	demands
encouraging, 103	concessions to, 153
eradication of, 165	enemy, 200
failure to understand, 119, 154, 246	groups, 106
fertile conditions for, 164	support of, 165
fighting, 131	havens, 184
financing	infrastructure, 104
disruption of, 159	leader, 256
follows, 3	organizations, 106, 182
global	disrupting and destroying, 159
as phenomenon, 155	terrorists, 117, 187, 197, 198, 200, 242,
sponsors of, 165	264
struggle against, 155	become criminals, 183
justifying, 164	become governments, 183
more, 116	defense against, 247
motivation for, 117	isolating, 158
mouvation for, 117	1501441115, 150

Palestinian	violence, 255
goals of, 156	tolerance, 98, 111
penetrate societies, 117	religious and ethnic, 145
striking deals with, 153	tolerant
trafficked with, 110	society, 181
trained, 158	toleration
test, 112	of ethnic
Thailand, 234	minorities, 181
theft, 52, 272	Tora Bora, 235
alleged by Osama bin Laden, 263	Torah, 260
them, 51	tort, 9
them-and-us	torture, 44, 75, 76, 265, 270
approach, 187	total
they and us, 153	control of life, 118
things	patriotism, 118
must change, 96	state control, 118
think	totalitarianism, 114, 137
freely, 240	trade, 211, 225
thinking	act 2002, 210
global, i	adjustment
Third Millennium, i	assistance, 213
threat	agreements, 210
assessment, 248	and development, 211
elimination, 203	and investment, 225
new, 198	free, 208
reduction, 197	government controlled, 208
to all nations, 194	in freedom, 129
threaten	lawfulness of, 208
neighbors, 189	liberalization, 210
threatening	preference
•	<u>*</u>
neighbors, 167	program, 213
threats, 44, 119	promotion
and enemies, 245	authority, 210
assessment of, 202	removal of restrictions, 209
common perspective on, 233	US-Russian, 237
fear of, 203	trade-unionism, 221
from terrorists and tyrants, 136	traditions, 221
health and environmental, 239	training, 232
immediacy of, 198	transatlantic
information on, 202	alliance of
inside the US, 248	democracies, 231
keeping pace with, 247	democracies, 233
response to, 231	transform
tit-for-tat	lives, 112

transformation	super-power, 116
of institutions	tyrants, 112
purpose of, 242	LINI 56 07 122 167
transforming	UN, 56, 87, 133, 167
intelligence	approval, 60
capabilities, 247	Conference
military	on Financing and Development,
structure, 245	218
transitional	endorsement, 60
safeguards, 213	framework
transnational	convention for international coop-
threats, 184, 185, 240	eration, 214
transparency, 91, 102, 240	inspections, 59, 60
traumatised	resolutions, 111
friend and ally, 234	disregard of by Israel, 271
treated as pawns, 111	Resolutions 242 and 338, 108
treaties, 11	sanctions
treatment	against Iraq, 194
and care, 226	use of
trial	by US, 271
put on, 75	unaccountable
tribal	few, 101
mindset, i	uncertainty, 245
rivalries, 143	of attack, 201
wars in Africa, 138	uncivilized
tribes, 7	nations, 156
true, 9, 34	understanding, 81
Truman	America, 250
administration, 166	informed, 149
trust, 195	unfair
public, 60	trade
trustee, 68	practices, 212
truth, 33, 147	trading
adherence to, 254	practices, 212, 213
existence of, 34	unforeseeable
honesty and justice, 275	events, 247
tuberculosis, 211, 226	unilateral
turban, 202	advantage, 115
two states	unipolar
meaning of, 86	sovereignty, 49
side by side, 85	unique
vision, 97	responsibilities, 252
tyranny, 8, 116 , 259, 270	United
tyrant	Europe
cjimiic	Larope

idea of 220	Australian
idea of, 230	alliance, 235
World, 230 unity, 76	bases and stations, 245
universities, 224	
unjustified	breach of agreement
attack, 69	in Qunduz, 271
,	cities, 251
unlawful combatants, 134, 264	Cold War
unlimited	apparatus, 186
powers	comprehensive
of nation states, 249	strategy, 209
UNO, 4, 11	contributions
unpredictability, 247	to global fund, 226
unsubstantiated	core value, 183
declaration, 255	defeat in Somalia, 261
unwanted	defense, 196
military	defense of, 244
presence, 272	diplomacy
US, 56, 60, 64, 67, 87, 102, 103, 115,	reliance on, 249
133, 176, 181, 201, 209, 211,	success, 249
213, 226, 236	diplomatic
access to	efforts, 249
distant theaters, 246	duplicity, 271
administration, i	in manners and values, 270
admittal	embassy
of defeat, 195	move to Jerusalem, 263
alliances	energy
in Asia, 235	waste of, 252
allies, 229	failure
and friends, 195	to understand terrorism, 251
armed forces, 246	forces, 245
reliance on, 249	abroad, 198
strength of, 245	overseas, 245
army	foreign
part of American people, 265	policy, 183, 189, 190, 270
as sponsor of terrorism, 165	foreign policy, 252
Asian	institutions, 186
alliances, 234	objectives, 272
friendships, 235	obstacle to secure world, 252
assets	freedom, 251
in outer space, 246	friends, 229
attack	funding of education, 227
on Afghanistan, 251	goal, 219
attempts	government, 99, 113, 201, 221
to justify war, 194	thinking, 187
J ,	<i>5</i> ,

hatred of, 192	population, 139
help, 227	portfolio
<u> -</u>	of military capabilities, 246
hypocrisy, 261, 271	• •
in manners and principles, 270 immunities	President, 246
	president, 76
for themselves, 271	priorities, 172
implementing	putting itself above law, 251
strategies, 228	reform
infrastructure, 246	agenda, 222
intelligence	refusal
advantages, 246	to provide proof, 230
interests	relations with
defense of, 249	Russia, 186
threatened, 232	relationship
international influence, 263	with Arab states, 172
international policy, 217	with China, 239
involvement	with Israel, 172
in Palestine, 256	resentment against, 247
with Arab States, 262	respect for
involvement in Chechnya, 261	rule of law, 250
key	responsibility
priorities, 225	for Israeli crimes against Pales-
leadership, 223, 224	tinians, 259
major strategies, 219	Russian
military	cooperation, 237
might, 198, 249	sanctions, 263
strength, 242	security and peace, 263
military bases, 263	stands for, 145
military threats, 263	state of diminished responsibility,
national	234
interests, 191	strength
security, 206, 252	and influence, 136, 255
security institutions, 242	foundation of, 252
security strategy, 225	success
nuclear bomb	reasons for, 242
on Japan, 270	support
occupation	for India in Kashmir, 272
alleged by Osama bin Laden, 263	for Israel, 247, 259, 272
outside of ICC jurisdiction, 250	for Israeli war, 107
partnership with India, 238	for Israeli war against Palestini-
policy	ans, 157
Netanyahu influence, 264	for Israelis over Jerusalem, 263
towards Russia, 237	for oppression of Palestinians,
poor record in morality, 227	265
poor record in morality, 221	203

systems of movement, 251	against enemies, 114
trade negotiations, 213	viable
victory	alternatives, 252
in Afghanistan, 195	vibrant economy, 102
vulnerabilities, 232	vicious
vulnerability, 251	circle, 243
war, 168	cycle, 82
in Afghanistan, 247	victims, 76
willingness	of injustice, 218
to use force, 245	victor justice, 76
WMD	victories
strategy, 195	and defeats, 244
us and them, 51	victory
US-China	easy, 246
relationship, 239	view
US-India	of the world
relationship, 238	distorted, 233
US-Russian	rational and informed, 183
differences, 237	vigilantes, 69
use of tanks	vilified
to suppress democracy	by media, 105
in Algeria, 270	vilify, 60
usurpation, 265	vilifying
usury, 268	opponents, 161
utopia	village-level
discredited, 137	elections, 240
Uygur	villain, 64
fighters, 157	violence, 89, 112, 228, 254, 262, 275
Uygurs	acts of, 82
repression of, 239	against innocents, 151
	and chaos, 121
vague	causes, 85
remarks, 86	encouraged
valid	by US, 179
logical conclusion	endless, 255
failure to draw, 151, 251	eruption of, 170
reason, 64, 71	excessive, 65
values, 111, 272	in self-defence, 255
alien	legitimate, 64
imposition of, 168	prevention of, 228
American, 140	restriction to minimum, 263
democratic, 167	subsidal of, 179
of freedom, 114	subsides, 107
protecting	to resolve disputes, 127

	: 4'6 1 62
unjustified, 64	justified, 63
use of, 63	major, 56
wanton, 65	never-ending, i
violent, 55	of global reach, 151
means	of ideas, 161
meaning of, 254	on terrorism, 239, 250
vision	ability to fight, 246
global, 113	parties to, 261, 262
two states, 97	prepare for, 228
visions	to save embarrassment, 59
of imminent attacks, 200	valid cause for, 61
voices	waging, 217
of the people, 174	will finish, 251
voluntary	winning, 252
cooperation, 142	zones
votes	regional, 184
attracting, 220	War on Terror, 49, 54, 96, 99, 100, 138,
	141, 157, 169, 190, 229, 234,
wage	247, 253, 256, 261, 262, 268
war, 76	alternatives to, 106
waging	doctrine, 105, 180, 183, 196, 255
wars, 75	Israeli use of, 174
wanton	fundamental flaw, 183
destruction, 200	hysteria, 162
war	Israeli use of, 97
acts of, 164	issues, 253
Afghan, 51, 230	opposition to, 256
against	paranoia, 248
terrorism, 235	use of
against a phenomenon, 151	by China, 239, 240
against global terrorism, 167	warfare
American, 76	new approaches, 246
casualties, 56	warlords, 49
caused by US, 251	warning, 254
crimes	wars, 55, 77, 115, 139, 144, 207, 254
by US, 271	and politics
criminals, 271	abandonment of, 265
erupts, 3	indulgence in by governments,
genocidal, 106	265
hysteria, 193	and terror
in Afghanistan, 246	decades of, 201
internal, 53	and terrorism, 171, 173
Iraq, 51	reality of, 265
justification, 60	lawless, 65
J	,

	1.6.1, 27
more, 116	default, 27
need to avoid, 56	draft, 28
nothing to be gained by, 199	final, 29
prevention of, 245	final by default, 29
reasons for, 245	summary, 26
removing causes of, 243	judges, 20
supporting, 75	juries, 20
terrorism and politics, 275	jurisdiction, 20
waging, 75	key people, 43
wanton, 69	location, 19
Washington, 100, 272	objections to draft judgement, 29
water, 224	operation, 20
use of, 227	parties in a case, 22
way	pleadings, 23
of life, 167, 221	powers, 20
way out	proof
quick and easy, 168	burden of, 24
WCJ, i, 81, 84	method of, 24
admittal	standard of, 25
of claim, 25	purpose, 19
appeals, 27	questionnaire, 85
challenge	refutations, 24
of claim, 25	relevance
challenging	of evidence, 24
decisions, 20	of facts, 23
facts, 24	reply, 26
claim, 25	role of, 11
clarification of issues, 28	stages of action, 22
commencement of actions, 22	statements of fact, 23
comments, 85	unresolved issues, 27
on evidence, 28	weakness
compared with ICJ, 37	Russian, 237
composition, 19	wealth
counterclaim, 26	distribution of, 219
court of last resort, 20	national, 125
defence, 25	natural creation and distribution, 218
denials, 24	weapons
dialogue between parties, 27	lead to means of defense, 243
entitlement to begin action, 21	of choice, 199
evidence, 23	of last resort, 199
examination of, 27	unconventional, 189
failure to produce, 27	wedding parties
judgement	bombed, 201
appeal against final, 29	bombings of, 166
,	5 /

1.1. 200	WD 50 50 55 450 400 404 405
weight, 208	WMD, 53, 60, 67, 160, 190, 194, 195,
welcome	200–202
intention, 113	bogus issue, 60
welfare, 99	destruction of, 197
well-being, 205	development, 67
well-meant	effects of use, 198
strategies, 219	elimination, 195
well-wisher, 266	enemies seeking, 119
West Bank, 86, 111, 175	in Israel, 271
Western	Iraq, 59, 67
Democracies, 161, 174, 220	possession, 67
Europe, 245	pretext for Iraq war, 271
Hemisphere, 182, 209, 214	prevent threats with, 142
hopes, 111	preventing
politicians, 220	spread of, 131
World, 9	prohibiting, 131
westernization, 161, 220	proliferation, 237
whims	spread of, 186
and fancies	•
	to be destroyed, 142
of politicians, 171	US policy on, 271
white	use of
race, 270	against US, 187
White House, 202	women
whole world	exploitation of
representation of, 256	in US, 269
widening	liberation of, 269
prosperity, 213	use of
widows, 260	in US, 269
Will	workers
Irresistible, i	find jobs, 223
will	protection, 213
and frivolity, 8	working
of majority, 148	democracy
of minority, 148	in Indonesia, 181
windmills, 183	World
wisdom, 229	Bank, 102, 176, 222, 224, 227
wishful	Court of Justice, see WCJ
thinking, 174, 179	Terrorism, 120
with us or against us, 105	world, 99
withdrawal	Americans ruling, 50
to pre-1967 borders, 177	domination, 195
withdrawals	economy
tactical, 179	strong, 204
witnesses, 208	free peaceful and secure, 238
	nee peacetal and secure, 250

global, 49	World Wars of 20th century, 119
inclusive	world-inclusive
institutions, 182	institutions, 233, 235
interconnected, 170	worship, 114
making better, 217	freely, 240
of coercion, 170	worst
of justice, 170	civilization, 268
order, 169, 172	wrong, 9
based on balance of power, 165	civil, 16
based on rule of law, 126	decision, 63
Netanyahu, 251	WTO, 133, 209, 211, 213, 237
secure, 143	China's entry, 240
peace and security, 237	requirements, 240
poor, 217, 223	•
rebuilding it, 217	Yehuda and Shomron, 175
safer place, 203, 234, 247	youth
single, 49	corrupted and mislead, 220
statesman, 99, 105	
wars, 208	Zionist
World War I, 11, 230	extremists, 99
World War II, 4, 11, 56, 101, 193, 217	Zionists, 164, 198